Lapap Mk II Manual

December 1, 2008

Lapap Mk II Manual

Edition Electronic version generated on Copyright © 2006 EADS SPACE Transportation

Contents

1	Inst	allatio	n Instructi	ons 3
-	11	Check	Platform	Requirements 3
	12	Instru	ictions for	Linux/Unix 3
	1.2	121	Unpacki	ng the Distribution 3
		1.2.1	Starting	the application 3
		1.2.2	Eviting	he application 3
	12	I.2.5	LATTING U	Windows YP 3
	1.5	121	Linna alci	willidows Ar
		1.3.1	Charling	ng me distribution
		1.3.2	Starting	the application
	1 4	1.3.3	Exiting t	
	1.4	Opera	ational Pro	ducts in the Standalone Version
2	Woi	king w	vith Lapap	Mk II 5
	2.1	Menu	Structure	5
		2.1.1	File	
			2.1.1.1	Reset Application
			2.1.1.2	Save Data pool Buffer
			2.1.1.3	Exit
		212	View	5
			2121	Show Message Buffer 5
			2.1.2.1	Refresh Display Navigation Tree
			2.1.2.2	Coogle / Table Style Soarch Regulte
			2.1.2.3	Google / Table Style Search Results 0 Zoom in dog / Zoom out dog 6
			2.1.2.4 2.1.2.5	Zoom in doc / Zoom out doc 6 Clear Lee 6
			2.1.2.5	Clear Log
			2.1.2.0	Windows
			2.1.2.7	WORKSpaces Image: Comparison of the second sec
			2.1.2.8	Add Plugin / Undock Plugin / Move Plugin / Remove Plugin 6
			2.1.2.9	Split Frame
			2.1.2.10	Maximize / Reset View
			2.1.2.11	Show Log Window
		2.1.3	Display .	
			2.1.3.1	Close
			2.1.3.2	Close Others
			2.1.3.3	Undock
			2.1.3.4	Cascade / Tile / Maximize All
			2.1.3.5	Toggle Tabbed Mode
			2.1.3.6	Synoptic Default Size
			2.1.3.7	Load Homepage
			2.1.3.8	Subsystem Overview
			2.1.3.9	Close all but Homepage
			21310	Find Text As You Type 7
			21311	Find Parameter In Other Displays 7
			21312	Properties 7
		214	Documer	ntation 7
		4.1.T	2141	Find in page
			2.1. 1 .1 2112	Zoom in dog / Zoom out dog
		21 ⊑	2.1.4.2	200mmuot / 200mout uot
		∠.1.3	21 = 1	Durafaran 20
		010	2.1.3.1 11.1	r references
		2.1.6	Help .	
			2.1.6.1	Hot Key Help 8
			2.1.6.2	Current Display
			2.1.6.3	ODF Documentation
			2.1.6.4	User Manual
			2.1.6.5	About

	2.2	Tool b	ar		. 8
	2.3	Works	paces .		. 10
		2.3.1	Plugins		. 10
		2.3.2	Predefir	ned Workspaces	. 10
		2.3.3	Changir	ng Workspaces	. 13
		2.3.4	Creating	g Workspaces	13
		235	Managi	ng Workspaces	17
		2.3.6	Croatin	a Ourn Windows	17
	2.4	2.5.0 Drofor	creating	g Own windows	10
	2.4	2 4 1	Andia N	······································	10
		2.4.1	Audio r		10
		2.4.2	Search	·····	18
		2.4.3	Datapoo	ol Buffer	. 18
		2.4.4	Tool Tip)S	. 19
•					
3	CLS	W Stat	us		21
	3.1	The Ta	abs of the	CLSW Browser	. 21
		3.1.1	Overvie	?W	. 21
		3.1.2	Port Co	unters	. 23
		3.1.3	Port Sta	tus	. 24
4	Syn	optic D	isplay Sy	ystem	25
	4.1	Windo	ow Hand	ling	. 25
		4.1.1	Docking	g / Undocking Displays	. 26
		4.1.2	Toggling	g Tabbed Mode	. 27
		4.1.3	Display	Size	. 27
			4.1.3.1	Changing Display Size	. 27
			4.1.3.2	Synoptic Default Size	. 27
	4.2	Synop	tic Navig	pation Buttons	27
	4.3	Displa	v Featur	ρς	28
	1.0	131	Ovorall	Monitoring Status	28
		4.3.1	Element		20
		4.3.2		Nercientien Bettene	20
			4.3.2.1		28
			4.3.2.2	Command Buttons / 2 Step Commanding	28
			4.3.2.3	Measurements	. 29
			4.3.2.4	Graphical Elements	. 29
		4.3.3	Display	Context Menus	. 29
			4.3.3.1	Context Menu for Buttons	. 30
			4.3.3.2	Context Menu for Labels	. 30
			4.3.3.3	Context Menu for Measurements	. 30
		4.3.4	Quick C	Graphs	. 32
			4.3.4.1	Purpose of Ouick Graphs	. 32
			4.3.4.2	Axis Functionality	33
			4343	Context Menu of a Quick Graph	33
			4344	Linggraph Monu	34
		135	T.J.T.T Porform	Diregraph Weitu	25
	4.4	4.5.5	Managan		25
	4.4		Charager	· · · · · · · · · · · · · · · · · · ·	
		4.4.1	Changi	ng Monitoring Values For Numerical Measurements	36
		4.4.2	Changii	ng Monitoring Values For State codes (Based on Discreet Measurements) .	37
5	Stat		lave		20
5	5 1	La Dist Station	nays Mode		20
	5.1	Station	Mada C	·····	37
	5.Z		i iviode C	companymity	40
	5.3 - ·		status	······································	40
	5.4	S-Band	a / KU-B	and	40

6	Svst	tem Me	essage Panel	41
	6.1	The M	lessage Buffer View	41
		6.1.1	The columns in the Message Buffer View	42
			6.1.1.1 Time	42
			6.1.1.2 Classification	42
			6.1.1.3 Subsystem	42
			6.1.1.4 Equipment	42
			6.1.1.5 Text	42
		6.1.2	Sorting the columns of the Message Buffer	43
		6.1.3	Actions in the Message Buffer View	43
			6131 Filter	43
			6132 Clear	45
			6133 Sync	45
			6134 Freeze	45
			6135 Live	45
			6136 Lord	45
			6.1.3.0 Load	45
			$\begin{array}{cccccccccccccccccccccccccccccccccccc$	43
			$6.1.3.6 \text{Acknowledge & Close} \dots \dots$	45
			6.1.3.9 Close	46
			6.1.3.10 Showing Event Help for a message	46
		6.1.4	The Icons in the Message Buffer View	46
			6.1.4.1 The "Live" Icon	46
			6.1.4.2 The "Loaded Messages" icon	46
			6.1.4.3 The "Filter" icon	46
7	C:L.	ation al	Axive#emoss	47
1	511u		Magazaga	47
	7.1	CAMUL		47
	7.2	SAW I		47
	7.3	Openi	Filter Experies and iter	4/
	7.4	SAW I		48
	7.5		A demonstration of CAMA management	48
		7.3.1		40
8	Tim	er Secti	ion	51
	8.1	Timer	Types	51
		8.1.1	Alarm Clock Timer	51
		8.1.2	Count Down Timer	52
		8.1.3	Timer Settings	52
			8.1.3.1 Name	52
			8132 Type	52
			8133 Sound	52
			0.1.0.5 Dound	52
9	Inte	rnation	al Procedure Viewer Components	53
	9.1	ODF E	Book Browser	53
		9.1.1	Selecting ODF Procedures	54
		9.1.2	Searching For Procedures	54
		9.1.3	Displaying Procedure Properties	56
	9.2	Procee	dure Executor	56
		921	Logic Flow Procedures	56
		, <u> </u>	9211 Working With Logic Flow Procedures	57
			9.2.1.1.1 Elements in Logic Flow Procedures	57
			9.2.1.1.2 Basic Navigation	59
			9.2.1.1.3 Executing The Logic Flow Procedures	59
		9.2.2	Check List Procedures	60
			9.2.2.1 Working With Check List Procedure	60
			9.2.2.1.1 Elements In Checklist Procedures	61
			9.2.2.1.2 Executing Check List Procedures	63
		9.2.3	PDF Procedures	63
		9.2.4	General Procedure Work Flow	63

	9.3 Documentation Viewer 9.3.1 Searching For Documentation	63 64
A	Keyboard Shortcuts and Hot Keys	65
B	Known Problems	67
	ndex	69

List of Tables

4	Synoptic Display System 4.1 Meaning of Measurements Background Color	29
5	Status Displays 5.1 Station Mode Values	39
A	Keyboard Shortcuts and Hot Keys A.1 Lapap Mk II Application and Desktop Hot Keys	65

Abstract

This document is the user manual for the Lapap Mk II Application.

Lapap Mk. II is the commanding and control crew interface for Columbus.

It provides online access to the nominal part of columbus' subsystems. Data is directly derived from the DMS data pool and commands are initiated directly to the onboard execution software by the crew.

Subsystems and equipment are shown within Lapap Mk. II as graphical schematics with data representation in alphanumerical output boxes, states, animated symbols and line color coding. Commands are predefined so that the crew can issue the commands by either clicking on a button or selecting the command from a list and clicking **Execute**.

Lapap Mk II provides commanding for all subsystems, display measurements acquired by the DMS and indicators of limit supervision and data validity.

Chapter 1

Installation Instructions

This chapter describes the steps necessary to install Lapap Mk II.

1.1 Check Platform Requirements

The Lapap Mk II software runs on Linux, Windows, and Solaris platforms for which a Java (TM) 2 Standard Edition runtime environment is available (see http://java.sun.com/j2se). It has been tested on the following platforms:

- SUSE Linux Enterprise Server 9 with Service Pack 3 and J2SE 1.6.0_01

- Windows XP (SP2), J2SE 1.6.0_01

Other platforms are expected to work but are currently not supported.

1.2 Instructions for Linux/Unix

1.2.1 Unpacking the Distribution

The distribution comes as an archive named *lapap-x.y.z.tar.gz* where *x.y.z* is the release version identifier (e.g., 1.4.0). To unpack this archive in a sub-directory of your home directory, do:

\$ cd \$HOME

\$ tar zxf lapap-x.y.z.tar.gz

This will create a directory *\$HOME/lapap-x.y.z* containing the complete Lapap Mk II distribution.

The Lapap Mk II software saves user specific information such as window position or workspace changes. They are saved into an application configuration directory which is by default named:

\$HOME/.lapap

If you install the Lapap Mk II software onto a machine where a previous version had been installed you may want to delete this directory or its contents to ensure a clean new environment and skip all former configuration changes.

1.2.2 Starting the application.

You can start the Lapap Mk II application with the command: \$ \$HOME/lapap-x.y.z/bin/lapap.sh

1.2.3 Exiting the application

Choose Exit from the File menu inside the application window.

1.3 Instructions for Windows XP

1.3.1 Unpacking the distribution.

The distribution comes as an archive named *lapap-x.y.z.zip* where x.y.z is the release version identifier (e.g., 1.4.0). To unpack this archive in a sub-directory of the program folder

- select the file *lapap-x.y.z.zip* in the windows explorer.

- in the context menu (right-click) choose **Extract All...** and extract the file to a directory of your choice, for example *C*:*Program Files*.

This will create a directory *C*:*Program Files**lapap-x.y.z* containing the complete Lapap Mk II distribution.

The Lapap Mk II software saves user specific information such as window position or workspace changes. They are saved into an application configuration directory which is by default named:

%USERPROFILE%/.lapap If you install the Lapap Mk II software onto a machine where a previous version had been installed you may want to delete this directory or its contents to ensure a clean new environment and skip all

1.3.2 Starting the application

former configuration changes.

You can start the Lapap Mk II application by double-clicking on the file: *C*:*Program Files**lapap-x.y.z**bin**lapap.bat*.

1.3.3 Exiting the application

Choose **Exit** from the **File** menu inside the application window.

1.4 Operational Products in the Standalone Version

In the standalone version of Lapap Mk II examples of operational data are contained for demonstration purposes. Operational products are Synoptic Displays, ODF Procedures and Crew Documentation. These products are located in dedicated sub-folders of the LAPAP Mk II directory structure:

share/displays

share/odf

share/doc

Here they can be easily exchanged if new data is available by simply replacing the files in these directories. However, Lapap Mk II expects all data in a consistent set and the paths to these data need to be preserved.

Chapter 2

Working with Lapap Mk II

This chapter describes the general work flow in Lapap Mk II. It will introduce the menu bar and the tool bar. Also information on working with **Workspaces** and **Plugins** can be found here.

2.1 Menu Structure

This section describes the different functions that can be found in the Menu of Lapap Mk II.

2.1.1 File

The File Menu contains functionality to save the data in the **Data pool buffer** and to exit the application.

2.1.1.1 Reset Application

Resets the application. This means that all read only workspaces are reloaded and changes performed to them are reverted.

In detail, the following actions are performed:

- Read only workspaces are reset to their initial layout.
- Documentation Viewer opens the default home page.
- Message Buffer View is set to its initial state.
- Preferences are set to their defaults. For detailed information about which settings are affected see Section 2.4.

2.1.1.2 Save Data pool Buffer...

The Data pool Buffer saves the values of the **Measurements** within the last 30 minutes and stores them automatically within a temporary binary file.

The menu item "Save Data pool Buffer..." is useful to export these data into *.csv format.

2.1.1.3 Exit

Choose this menu item to quit the Lapap MK II application. Before the application will be closed, a confirm dialog appears which has to be confirmed by clicking the **OK** button.

2.1.2 View

This menu contains functionality to manage the different views of the application.

2.1.2.1 Show Message Buffer ...

Choose this menu item to open the **Message Buffer View**. For detailed information about the **Message Buffer View** see Chapter 6.

2.1.2.2 Refresh Display Navigation Tree

Refreshes the Synoptics Hierarchy View.

2.1.2.3 Google / Table Style Search Results

These two menu items change the layout of the search results in the documentation workspace. **Google Style Search Results** will make the results look like the results in *www.google.com*, while **Table Style Search Results** displays the search results in a table.

2.1.2.4 Zoom in doc / Zoom out doc

This functionality changes the font size of the content in the documentation workspace. Images are not affected by the zoom level and will stay in their default size (100%).

2.1.2.5 Clear Log

Clears the Lapap Mk II applications log. This log displays messages which track the user actions performed in Lapap Mk II.

2.1.2.6 Windows

The **Windows** sub menu contains the menu item: **New Window...** and a list of all windows which have been created for the currently selected **Workspace**. The functionality of configuring **Workspace**s and new windows is described in Section 2.3.

2.1.2.7 Workspaces

Opens a sub menu to choose a different **Workspace**. **Workspaces** can be seen as different views of the content available in Lapap Mk II.

For detailed information see Section 2.3.

2.1.2.8 Add Plugin / Undock Plugin / Move Plugin / Remove Plugin

Lapap Mk II offers the possibility to load and integrate special **Plugins** into the **Workspace**. It is also possible to undock the currently selected view, which will open it in an external window.

2.1.2.9 Split Frame

By using this menu item it is possible to split a frame into two new Frames. See Section 2.3 for detailed information on Workspace layout.

2.1.2.10 Maximize... / Reset View

Maximizes the content of the active frame in the working area. After using this feature, the menu item is renamed **Reset View** and will reset the view.

2.1.2.11 Show Log Window

Displays the applications user log. In this area near the bottom of the application window user actions will be logged.

2.1.3 Display

2.1.3.1 Close

Closes the currently selected Display.

2.1.3.2 Close Others

Closes all other Displays except the one currently selected.

2.1.3.3 Undock

Undocks the current Display. See Section 4.1.1 for more information.

2.1.3.4 Cascade / Tile / Maximize All...

These are default window operations for the open displays.

IMPORTANT

 \rightarrow The Application has to be in Window Mode to make these functions work.

See Section 2.1.3.5.

2.1.3.5 Toggle Tabbed Mode

Toggles the display mode of Synoptic Displays. See Section 4.1.2 for more information

2.1.3.6 Synoptic Default Size

Resizes a Display back to its native size. See Section 4.1.3.2.

2.1.3.7 Load Homepage...

Opens the Home Display.

2.1.3.8 Subsystem Overview

Opens the Functional Overview Displays of all subsystems.

2.1.3.9 Close all but Homepage

Closes all open displays except the **Home Display**.

2.1.3.10 Find Text As You Type

This functionality is related to **Synoptic Displays**. It can be used to find a specific item within the display by searching for its name as it is shown in the display.

If a display is open, select **Find Text As You Type** and type the term you are searching for. The text you have entered so far is displayed in the lower section of the display area. All items within the display which match to the search term are highlighted with a yellow frame.

To stop the function press the **ESC** key.

2.1.3.11 Find Parameter In Other Displays

This function can be used to find other displays that contain a certain **Measurement**. See Find Parameter In Other Displays.

2.1.3.12 Properties

Opens a dialog which shows the properties of the selected Display.

2.1.4 Documentation

The functionality found here is related to the Documentation Workspace.

2.1.4.1 Find in page...

This will open a small dialog box to enter a search term. The currently loaded documentation page will then be searched for the entered term.

2.1.4.2 Zoom in doc / Zoom out doc

This will aply a larger / smaller font size to the content in the documentation frame. Images are not affected by the selected size. Alternatively this functionality can be accessed from the lapap tool bar.

2.1.5 Options

This menu contains only one entry: The **Preferences** Dialog.

2.1.5.1 Preferences

Opens the **Preferences**dialog of Lapap Mk II. See Section 2.4.

2.1.6 Help

2.1.6.1 Hot Key Help

This will open a help screen which gives information about certain hot keys (key combinations) and about their functionality.

2.1.6.2 Current Display

This menu item opens the display specific help. Every display contains a link to a specific help page which will be displayed in the **Documentation Workspace**.

2.1.6.3 ODF Documentation

Opens the ODF Documentation in the **Documentation Workspace**.

2.1.6.4 User Manual

Displays this document in the Documentation Workspace.

2.1.6.5 About

Opens an **About** dialog. The exact version and build number of the Lapap Mk II application can be seen here.

2.2 Tool bar

This chapter gives an introduction to the Lapap tool bar.

🛲 lapap 🎐	
<u>File View Display Documentation</u>	Options Help
💽 😳 🔏 Synoptics (R)	
💻 🍕 🕖 🌮 🖇 🕼 sw	S DMS S Ku
COL Synoptics Home Page GOL Hot Key Table GOL Symbols	DMS Monitoring O DMS Monitoring O DMS Monitoring DMS Monitoring
📃 — 📰 COL System SW Overview	

Tool bar of Lapap Mk II

• Hot Key Help

This will open the Hot Key Help. "Hot Keys" are keyboard shortcuts which provide a fast way to reach certain functionality within the application.

🛿 Hot Key Help <3> 🅘		_ 🗆 X
Description	Hot Key	
Perform dialog default action Activate selected element Cancel selection	Enter Space ESC	<u> </u>
Activate window menu Close active window Move active window Resize active window Minimize active window Maximize active window	Alt + F3 Alt + F4 Alt + F7 Alt + F8 Alt + F9 Alt + F10	=
Activate next window Activate previous window Toggle desktop Popup launch menu Show window list Mouse emulation using →, -, 1 and 1 Window screenshot Desktop screenshot Logout Show task manager	Alt + Tab Alt + Shift + Tab Ctrl + Alt + D Alt + F1 Alt + F5 Alt + F12 Alt + Print Ctrl + Print Ctrl + Alt + Delete Ctrl + ESC	
Show Klipper popup-menu Manually invoke action on clipboard Enable/disable clipboard actions	Ctrl + Alt + V Ctrl + Alt + R Ctrl + Alt + X	•
C	lose	

Hotkey Help Dialog

• Show Message Buffer

This will open the Message Buffer View. See Section 6.1.

• Workspace Chooser

Synoptics (R) This element can be used to choose a predefined Workspace. See Section 2.3.

• Maximize Frame

This will maximize the currently selected frame in the workspace. See Section 2.1.2.10.

• Window Actions

These Actions are used to control certain functions of the Synoptic Display Windows. For more Information on how to handle windows in lapap see Section 4.1.

The actions are (from left to right):

- Cascade. Organizes the currently opened displays in a cascaded way.
- Tile. Organizes the currently opened displays in a tiled way.
- Maximize Maximizes the currently selected display within the Synoptic Display Area
- **Toggle Tabbed Mode**. This toggles between the tabbed and the windowed mode. See Section 2.1.3.5.

• Synoptic Actions

These Actions are used to invoke certain actions specific for Synoptic Displays. See Chapter 4.

The actions are (from left to right):

- **Default Synoptic Size**. Sets the currently selected display to its Synoptic Default Size. See Section 2.1.3.6.
- Load Homepage. This opens the Home Display.
- Subsystem Overview. This opens the Functional Subsystem Displays for all subsystems.
- Close all but Homepage. All other displays than the Homepage will be closed.

• Zoom Documentation

This will zoom in / out in the Section 9.3 **Workspace**. Zooming the documentation means that the font size of the documentation content will be increased / decreased. Images are not affected by the zoom level.

CAUTION

When changing the font size of the documentation the currently visible position of the document is not maintained.

• Reset Application

Resets the application. This means that all read only workspaces are reloaded and changes performed to them are reverted.

• Back List / Forward List

These controls are related to **Synoptic Procedures**. They allow the navigation through the history of procedures which have been opened in the **Procedure Executor**. Threre are two ways to use the controls:

A simple click will move to the previous / next procedure in the history

A click on the small arrow on the left of the control will open a popup menu displaying all procedures that lie before / behind the currently selected procedure. To open a procedure from this list, simply click it.

2.3 Workspaces

Workspaces can be considered as different display representations of the functionality and tools available in Lapap Mk II.

Workspace defines the frames presenting the crew interface to the Lapap Mk II tools. Frame locations are fixed whereas the occupancy is defined by different frame sets, i.e. **Workspace**s. The menu bar and the tool bars are not part of a **Workspace** and always present.

The predefined **Workspaces** which can be selected in the view Menu, but it is also possible to define own **Workspaces**.

2.3.1 Plugins

Workspaces display their content using **Plugins**. A **Plugin** can be considered as a component that can be integrated dynamically into the application. In fact, everything (except the menu and the tool bars) within lapap is realised as **Plugin**s.

As these **Plugin**s are realized as self contained functional units, it is possible to customize the layout of the application.

2.3.2 Predefined Workspaces

The following **Workspace**s are predefined and can be selected in the Menu **View** \ **Workspace**s. It is possible to select a **Workspace** from the **Workspace** Chooser in the Lapap tool bar. Each **Workspace** is designed for a special purpose.

One thing that all predefined **Workspace**s have in common, is that the System Message Panel is always present.

• Synoptics

This is the default **Workspace** loaded at start up. This **Workspace** should be used to work with the **Synoptic Displays** exclusively.

🔤 lapap 🎐						×
File View Display Documentation Options	Help					
Cesa 💰 Synoptics (R)	3 6880000000000	م م 🧛 🖕 🔸 🔸				Display Refresh 😐 GPS 14Jun07 10:13:52
💻 🍕 🕖 🦨 🕼 🕼 sw	🕼 DMS S Ки			Timer 1 00:00:00 -		ner 3 0:00:00 + 0 00:00:00 + 0 00:00 + 0 00:00 + 0 00:00 + 0 00:00 + 0 00:00 + 0 00:00 + 0 00:00 + 0 00:00 + 0 00:00
COL Synoptics Home Page	DMS Monitoring OMS	Monitoring 🛛 👄 DMS Monitoring	🖗 COL System SW 0 🌾 \varTheta Mon	itoring CTCU 🔴 COL Synoptic:	H	
COL Symbols COL Symbols COL System SW Overview MTL Status		DMS Monitoring		1	DMS Monitoring	(ersion: Ops Products 3.1.0
Get Cabin Temp Get Cabi	External				Ovhd Fwd	
► I DMS	Payload					Node 2
← C ECLSS		CMU 2 04	03	ISPR 02 ETC	ISPR 01 FSL	
🗠 🔚 EPDS	SOLAK	CMU 3 Storage	Storage	Pwr	Pwr	
• = TCS	EPF SQZ	MMC		Main Pwr Bus Off	fain Pwr Bus Off	Set
← ☐ Payload ← ☐ SIVQ	N Off			Aux Pwr Bus Off /	Aux Pwr Bus Off	Cabin Temp
	R Off	1			Ovhd Aft	
ECLSS Search	EulEr Panel 2	CLSW 2	IRE 2 IFOR AS LEPM	KIRD AD E BIOLAB	FOR ALL ER 6	<u> </u>
ODF Libraries	SOX XCMU	CMU 4	WT PWT	Pwr	PW	
COMMS	N Off ATU 2	MMU 1 Main Pwr Bus 0	ff Main Pwr Bus Off	Main Pwr Bus Off	tain Pwr Bus Off	
2.305 - VCRs Activation/Deactivation	R Off VMN 1	Spare Aux Pwr Bus	ff Aux Pwr Bus Off	Aux Pwr Bus Off	Aux Pwr Bus Off	Environment
- DMS	VMN 2			VII	011	ECLSS
Activation and Checkout 1 201 - CMU 1 (2.3.4) Activation and	VCR 1				5UD 1 40 D 1	Cabin Temp1
- 1.202 - PLCU Activation and Checkout	EPF VCP 2		SUP 2		Aft Deck	СТСИ1
- 1.203 - XCMU Activation and Checkou - 1.211 - CMU 1 (2,3,4) Deactivation	SDX VCK 2			100		22.0 degC
- 1.212 - PLCU Deactivation 1.213 - XCMU Deactivation	N Off	HRM Storage	S/S CTCU 2	S/S CTCU 1	CHX 1/2	СТСU2
- Corrective	R Off PPSB	PLCU	PDU 2	PDU 1	CWSA 1/2	M degC
PECES P SIVQ		VDPU	VTC 2	VTC 1	WPA 1/2	TPS1 Air Press
P DMS P Malfunction	EPF					750.1 mmHg
- 3.220 - PLCU Failure - PLCU - 3.252 - Payload Bus Failure - PLCU - 3.253 - Payload Bus FT Status Fro	SDN	SUP 4		SUP 3	Fwd Deck	1.5 mmHg
- Corrective	R Off	CLOW 1	_			PPOS1 Press
- EPDS		ISPR F4	IRF 1 ISPR F3 MARES	ISPR F2 MSG	ISPR F1 EDR	157.5 mmHg
· 16	PPSB Exp	емот	wr Pwr	Pwr	Pwr	CAW + APP
	Pwr Stat	DMC Main Pwr Bus 0	ff Main Pwr Bus Off	Main Pwr Bus Off	Iain Pwr Bus Off	Panel 1
	R: Redun	MMU 2 Aux Pwr Bus 0	ff Aux Pwr Bus Off	Aux Pwr Bus Off	Aux Pwr Bus Off	Activation
						Part I
Activating /col_data/odf/DMS/ACTIVATION & C						
Time Classification Subsystem	n Equipment	Incolary (DRIORTICS, ROOT (COL, HOM	Euro frazion	Text		
14jun07 10:13:34 ERROR 14jun07 10:10:56 ERROR	Laptop3 Display /col_data/c	lisplays/SYNOPTICS_ROOT/COL_HOMI lisplays/SYNOPTICS/MONITORING/MC	Luss frozen			
14Jun07 10:08:56 ERROR	Laptop3 Display Monitoring	CTCU1_Cabin_Temp1_DMC frozen				
14jun07 10:08:13 ERROR	Laptop3 Display Monitoring	CTCU1_Cabin_Temp1_DMC frozen				

Synoptics Workspace

• Synoptics & Proc.

This Workspace makes it possible to work with both Procedures and Synoptics at the same time.



Synoptics & Proc. Workspace

• Procedures

This Workspace focuses on working with procedures.

File View Display Documentation Options Help		
Cesa 2 Procedures (R)	2 8 8 10 4 8 10 10 10 10 10 10 10 10 10 10 10 10 10	Display Refresh 🖨 GPS 14Jun07 10:15:24
	Ku Timer 1	3 Timer 2 3 Timer 3 3 Timer 4 3
ECISS Search	1.201 CMU 1 (2.3.4) ACTIVATION AND CHECKOUT	
ODF Libraries	Hide Parameters Show Ground Blocks	
Provinal		
2.305 - YCRs Activation/Deactivation	Select CMU GENERIC 1 2 3 4	
P DMS		
 Activation and Checkout 1.201 CMU 1 (2.2.4) Activation and Checkout 	ABICCTVC.	
- 1.202 - PLCU Activation and Checkout	This procedure shall be used to artivate CMU 1.2.3 or 4 during DMS Nominal Mode	
- 1.203 - XCMU Activation and Checkout		
- 1.212 - PLCU Deactivation	NOTE	
- 1.213 - XCMU Deactivation	 If activation of nominal CMU1(2,3,4) is not possible the redundant CMU1(2,3,4) will be activ 	ated by the FLAP.
► ECLSS	 An active nominal DMS is precondition for CMU1(2,3,4) activation. 	
e sive	 Exclusively nominal or redundant branch of CMU1(2,3,4)shall be active. 	
P - Malfunction		
3.220 - PLCU Fallure - PLCU	 VERIFY NO BRANCH OF CMU1(2,3,4) S ACTIVE 	
3.252 - Payload Bus Fallure - PLCU 3.253 - Payload Bus RT Status Error	PWS DWS: CMU10.3.4)	
 Corrective 	CMU1(2.3.4)	
ECLSS FPDS		
◆ TCS	Verify Nom Pwr A - OFF	
	Verify Nom Pwr B - OFF	
	Verify Reduin Par R - OFF	
	[GROUND BLOCK HIDDEN]	
	2. ACTIVATE CMU1(2.3.4)	
	PWS DMS: CMU1(2,3,4): CMU1(2,3,4) Commands	
	CM U 1(2,3,4) Commands	
	'Commands'	
	cmd Activation Execute (OK)	
	Wait 60 seconds before verification	
	If Nom Ready Stat - READY (green background)	
	-/hom Parr A - Ohl	
	√Nom Pwr B - ON	
	√Redun Pwr A - 0FF	
	√Redun Pwr B - OFF	
	Verify Nom Unit - AVAIL	
	lf Redun Ready Stat – READY (green background)	
	√Redun Pwr A - ON	
	√Redun Pwr B - ON	
	√Nom Pwr A - OFF	
	Vom Pwr B - OFF	
Activating /col_data/odf/DMS/ACTIVATION & CHECKOUT/1.201_E_D.	Yenry nom onn - onwwar	×
Time Classification Subsystem Equipment	Text play (rol. data/displays/SYNOPTICS_ROOT/COL_HOME uss frozen	
14jun07 10:10:56 ERROR Laptop3 Dis	play /col_data/displays/SYNOPTICS/MONITORING/MON_1.uss frozen	
14Jun07 10:08:56 ERROR Laptop3 Dis	olay Monitoring CTCU1_Cabin_Temp1_DMC frozen	
14jun07 10:08:46 ERROR Laptop3 Dis 14jun07 10:08:12 ERROR Laptop3 Dis	olay Monitoring CTCU1, Cabin, Temp1, DMC frozen	
Ladinois accoras la const	and according of conformation and a found and and a second s	

Procedures Workspace

• CLSW Status

This workspace can be used to access the web front ends of the four Columbus CLSW Switches. See Chapter 3 for more information.

Elle Yi	pap ew <u>D</u> isp) ay Qpt	ions <u>H</u> el	lp																								_ 🗆 X
Cost) 💰 🖸	LSW Sta	tus (R)	-	8 6	980) (8	19 0	< 6	ب	•	÷ •													Display	Refresh \varTheta 13Fe	b07 12:54:39
<mark>묘</mark> 4	l 🦉 ;	\$	健 sw		Ì	DMS	s H	(u												Time 00:0	r 1 0:00 :	• • •	Timer 00:00	2		Timer 3 00:00:00	Timer 4 00:00:0	
als C	LSW 10 S	Status	3 CL	SW 11 Statu	15 3	🕴 CLSW	20 Sta	tus	🕸 CLSV	V 2.1 Stat	us																	
Overv	iew							Port Co	unters							Port Sta	tus											
4.0%												1	Port Util	lization													Legend	Tx
25%																											5 Non-Unicast Pkt	s Rx
1.0%		_		-																							Port Connected	
	0	é	ó	ò	ò	ò	ó	ő	ò	0	0	0	0	0	0	0	ő	0	0	0	0	Ő	0	0	ø	ø	ØPort Disabled	•
Status	Alert	of link		Date / T	ime	45-05-5	D.M.	Descripti	on	multiple	desires		2															10
	Loss	or link		FED 13, .	2007 12:	40:20 F	rm	Lost conf	lection to	multiple	devices	on port.	2.															
Refre	ish																						Open Ev	ent /	cknowle	dge Selected Ev	ents Delete Se	lected Events
	Time	Cla	ssification	1 Subsyst	tem Eq	ulpmen	ıt												Text									
13Feb0 13Feb0	7 12:53: 7 12:53:	34 INFO 29 INFO		DMS DMS	PWS PWS	3	Link Loss	to DMS e of data I	stablishe ink to DM	d: Comn (S: No co	nand and Immand	d monitor and mon	ring capa litoring c	ability for apability	this PWS from this	is availai s PWS ava	ble. Blable.											
13Feb0 12Feb0	7 10:04:	55 INFO 09 INFO		ECLSS DMS	CTC	U1 3	USM. Link	SW_DMC	_USS_Swi stablishe	op_Instar d: Comn	nce : Mor nand and	nitoring S d monitor	ervice: C ing capa	TCU1_C	abin_Ten this PWS	is availa	back int ble.	o Soft-Li	mit. Activ	e Set 0.	Currient \	/alue = :	19.88.					
12Feb0	7 15:47:	15 INFO		DMS	PWS	3	Loss	of data I	ink to DM	IS: No co	immand	and mon	litoring c	apability	from this	s PWS ava	ilable.											

CLSW Configuration Workspace

• Documentation

The focus of this Workspace is to display the Documentation available in Lapap Mk II.

an Iabab 🤤					X
File View Display Documentation C) <u>p</u> tions <u>H</u> elp				
Cesa 3 Decumentation (R)		1 * 🟠 🗞 🕾 🔍 🗛 🗛 🗛 🖕 🚽 🚽			Display Refresh
🖳 🍕 🕖 🎉 🔒 sw	DMS S	Ku	Timer 1	Timer 2 00:00:00	Timer 3 Timer 4 00:00:00
EC122	Search	Back Forward			
Search: Procedures Documentat	ion 🔾 Displays 🔾 All	Crumb Trail: <u>RefDoc</u> > LAPAP Doc Plugin Test			
Search & Tales O Contents O Bot I result(1) found for search ECL (Incestati Proceeding 3.01 COLECL TABLES COLECUS Pranement Table : Calin har Ment Cole Cole Stramener Table : Calin har Ment Cole Cole Stramener Table : Calin har Ment Cole : Cole : Calin Cole : Calin Cole (Calin Cole : Cole : Calin Cole : Calin Cole Cole : Cole : Calin Cole : Cali	S 557 PARAMETER Assembly (77A) Assembly (77A) 	 Compare the term of t			
Time Classification St	ubsystem Equipment	Text			
14Jun07 10:10:56 ERROR	Laptop3 D	isplay /col_data/displays/SYNOPTICS/MONITORING/MON_1.uss frozen			
14Jun07 10:08:56 ERROR	Laptop3 D	isplay Monitoring CTCU1_Cabin_Temp1_DMC frozen			
14Jun07 10:08:46 ERROR	Laptop3 D	isplay Monitoring CTCU1_Cabin_Temp1_DMC frozen			
14jun0/ 10:08:13 ERROR	Laptop3 D	ispiay Monitoring CICUL Cabin, Temp1, DMC frozen			
14jun0/10:08:0/ ERROR	Laptop3 D	Isplay Monitoring CICU1_Cabin_Temp1_UMC trozen			

Documentation Workspace

2.3.3 Changing Workspaces

It is possible to change the layout of a **Workspace** (predefined or self created) and add or remove **Plugins**. If the layout of a **Workspace** has been changed and the application is closed, this changes will be saved and restored at the next start of lapap.

Predifined **Workspace**s marked with an **(R)** are treated in a special way. The **(R)** stands for "read only". These read only **Workspace**s can be changed in layout and content just like every non read only **Workspace** but these changes aren't saved when the application is closed. Another difference is that the original layout of read only **Workspace**s will be restored if the **Reset Application** icon in the toolbar is pressed.

2.3.4 Creating Workspaces

To create an empty **Workspace**, choose **View** \ **Workspace** \ **Create Workspace**... from the menu. This will create a new **Workspace** and remove all **Plugins** from the application leaving an empty working area.

At first, the user is requested to enter a name for the new Workspace in the upcoming dialog box.

Impo	RTANT
	The menu item view \ Workspace \ Create Workspace will create a new Workspace in the application main window. For reasons of readability the following figures will show the process of configuring an own workspace in a new window. See Section 2.3.6.
	Meil Enter workspace name Image: Constraint of the second secon

Creating a new Workspace

After pressing the **OK** button, the new empty **Workspace** will be selected.



IMPORTANT

The steps described in the following can also be performed on the predefined Workspaces. But changes performed on Workspaces marked with an (R) won't be saved when the application is closed.

Now it is up to the user to design a new **Workspace**. It is possible to subdivide the working area into smaller panels which are divided by horizontal or vertical splitters.

This can be done by right clicking anywhere into the empty frame and selecting **Split Frame** from the context menu. This will divide the frame in two by a horizontal splitter. The orientation of the sliders can be changed by context menu. To remove a frame set, choose **Remove Frameset** from the splitter's context menu. Of course, a frame of a frame set can be divided into two more frames



1: Context Menu in an Empty Frame



2: The Empty Frame is Split into a Frame Set



4: Frame Set After Changing the Orientation The context menu for a splitter contains these menu items:

Change Orientation

This will change the splitters orientation from vertical to horizontal and vice versa.

• Remove Frameset

This removes the splitter and unites the two frames into one. The **Plugin**s which have been integrated into the two frames are now all integrated into one Frame.

• Resize Frameset

This will open a sub menu with radio buttons to choose the way the frameset should behave if the application window is resized or the splitter is moved. Only one of the following settings can be selected:

- Proportional

The Frames will be resized proportional, if the application window is resized. The splitter can be moved manually.

- Proportional, Fixed

The Frames will be resized proportional, if the application window is resized., The splitter can't be moved manually.

– Left / Top

The left / top Frame will be resized proportional while the right / bottom Frame keeps its size. The splitter can be moved manually.

- Left / Top, Fixed

The left / top Frame will be resized proportional while the right / bottom Frame keeps its size. The splitter can't be moved manually.

- Right / Bottom

The right / bottom Frame will be resized proportional while the top / left Frame keeps its size. The splitter can be moved manually.

- Right / Bottom, Fixed

The right / bottom Frame will be resized proportional while the top / left Frame keeps its size. The splitter can't be moved manually.

After the working area has been organized into frame sets, it is time to put **Plugin**s into them:

- Choose **Add Plugin** from the context menu. This will bring up a list of **Plugins** at the right side of the applications window.
- Just drag and drop the **Plugins** into their destination frame.

🕷 lapap -	Myworks	pace 🥞 🔤	- □ >
			1
	Add Plugir	Columbus LAN Switch Status	
	Split Fram	e Display Hierarchy	
		ODE Book Browser	
		ODE Procedures	
		ODE Search	
		Synoptic Displays	
		System Message Panel	1
			. .
: Add	Plugi	n Opens the Plugir	ns Li
: Add	l Plugii MyWorks	n Opens the Plugir	ns Li
: Add	Plugii MyWorks	n Opens the Plugir	ns Li . □ >
: Add	I Plugii MyWorks	n Opens the Plugir	ns Li
: Add	Plugii MyWorks	n Opens the Plugin	
	Diagram Plugit		
Add	Plugii MyWorksi	n Opens the Plugin pace	
Add	MyWorks		
: Add Iapap - Col s Col s	Plugin MyWorks Mymoptics H Max 2		
COLS	Plugin MyWorks Synoptics H Synoptics H Sin Difference Sin Difference <t< td=""><td></td><td></td></t<>		
Cols Cols Cols Cols Cols Cols Cols Cols Cols Cols Cols Cols Cols Cols Cols Cols Cols Cols Cols Cols Cols Cols Cols Cols Cols Cols Cols Cols Cols Cols Cols Cols Cols Cols Cols Cols Cols Cols Cols Cols Cols Cols Cols Cols Cols Cols Cols Cols Cols Cols Cols Cols Cols Cols Cols Cols Cols Cols Cols Cols Cols Cols Cols Cols Cols Cols Cols Cols Cols Cols Cols Cols Cols Cols Cols Cols Cols Cols Cols Cols Cols Cols Cols Cols Cols Cols Cols Cols Cols Cols Cols Cols Cols Cols Cols Cols Cols Cols Cols Cols Cols Cols Cols Cols Cols Cols Cols Cols Cols Cols Cols Cols Cols Cols Cols Cols Cols Cols Cols Cols Cols Cols Cols Cols Cols Cols Cols Cols Cols Cols Cols Cols Cols Cols Cols Cols Cols Cols Cols Cols Cols Cols Cols Cols Cols Cols Cols Cols Cols Cols Cols Cols Cols Cols Cols Cols Cols Cols Cols Cols Cols Cols Cols Cols Cols Cols Cols Cols Cols Cols Cols Cols Cols Cols Cols Cols Cols Cols Cols Cols Cols Cols Cols Cols Cols Cols Cols Cols Cols Cols Cols Cols Cols Cols Cols Cols Cols Cols Cols Cols Cols Cols Cols Cols Cols Cols Cols Cols Cols Cols Cols Cols Cols Cols Cols Cols Cols Cols Cols Cols Cols Cols Cols Cols Cols Cols Cols Cols Cols Cols Cols Cols Cols Cols Cols Cols Cols Cols Cols Cols Cols Cols Cols Cols Cols Cols Cols Cols Cols Cols Cols Cols Cols Cols Cols Cols Cols Cols Cols Cols Cols Cols Cols Cols Cols Cols Cols Cols Cols Cols Cols Cols Cols Cols Cols Cols Cols Cols Cols Cols Cols Cols Cols Cols Cols Cols Cols Cols Cols Cols Cols Cols Cols Cols Cols Cols Cols Cols Cols Cols Cols Cols Cols Cols Cols Cols Cols Cols Cols Cols Cols Cols Cols Cols Cols Cols Cols Cols Cols Cols Cols Cols Cols Cols Cols Cols Cols Cols Cols Cols Cols Cols Cols Cols Cols Cols Cols Cols Cols Cols Cols	MyWorks MyWorks Synoptics H My My My My My My My My My My		
Add I lapap - Cols Cols Cols Cols Cols Cols Cols Cols Cols Cols Cols Cols Cols Cols Cols Cols Cols Cols Cols Cols Cols Cols Cols Cols Cols Cols Cols Cols Cols Cols Cols Cols Cols Cols Cols Cols Cols Cols Cols Cols Cols Cols Cols Cols Cols Cols Cols Cols Cols Cols Cols Cols Cols Cols Cols Cols Cols Cols Cols Cols Cols Cols Cols Cols Cols Cols Cols Cols Cols Cols Cols Cols Cols Cols Cols Cols Cols Cols Cols Cols Cols Cols Cols Cols Cols Cols Cols Cols Cols Cols Cols Cols Cols Cols Cols Cols Cols Cols Cols Cols Cols Cols Cols Cols Cols Cols Cols Cols Cols Cols Cols Cols Cols Cols Cols Cols Cols Cols Cols Cols Cols Cols Cols Cols Cols Cols Cols Cols Cols Cols Cols Cols Cols Cols Cols Cols Cols Cols Cols Cols Cols Cols Cols Cols Cols Cols Cols Cols Cols Cols Cols Cols Cols Cols Cols Cols Cols Cols Cols Cols Cols Cols Cols Cols Cols Cols Cols Cols Cols Cols Cols Cols Cols Cols Cols Cols Cols Cols Cols Cols Cols Cols Cols Cols Cols Cols Cols Cols Cols Cols Cols Cols Cols Cols Cols Cols Cols Cols Cols Cols Cols Cols Cols Cols Cols Cols Cols Cols Cols Cols Cols Cols Cols Cols Cols Cols Cols Cols Cols Cols Cols Cols Cols Cols Cols Cols Cols Cols Cols Cols Cols Cols Cols Cols Cols Cols Cols Cols Cols Cols Cols Cols Cols Cols Cols Cols Cols Cols Cols Cols Cols Cols Cols Cols Cols Cols Cols Cols Cols Cols Cols Cols Cols Cols Cols Cols Cols Cols Cols Cols Cols Cols Cols Cols Cols Cols Cols Cols Cols Cols Cols Cols Cols Cols Cols Cols Cols Cols Cols Cols Cols Cols Cols Cols Cols Cols Cols Cols Cols Cols Cols Cols Cols Cols Cols Cols Cols Cols Cols Cols Cols Cols Cols Cols Cols Cols Cols Cols Cols Cols Cols Cols Cols Cols Cols Cols Cols Cols Cols Cols Cols Cols Cols Cols Cols Cols Cols Cols Co	Myworks Mynoptics H Mine Mine Mine Mine Mine Mine Mine Mine		
Constant Constant Constant Constant Constant Constant Constant Constant Constant Constant Constant Constant Constant Constant Constant Constant Constant Constant Constant Constant Constant Constant Constant Constant Constant Constant Constant Constant Constant Constant Constant Constant Constant Constant Constant Constant Constant Constant Constant Constant Constant Constant Constant Constant Constant Constant Constant Constant Constant Constant Constant Constant Constant Constant Constant Constant Constant Constant Constant Constant Constant Constant Constant Constant Constant Constant Constant Constant Constant Constant Constant Constant Constant Constant Constant Constant Constant Constant Constant Constant Constant Constant Constant Constant Constant Constant Constant Constant Constant Constant Constant Constant Constant Constant Constant Constant Constant Constant Constant Constant Constant Constant Constant Constant Constant Constant Constant Constant Constant Constant Constant Constant Constant Constant Constant Constant Constant Constant Constant Constant Constant Constant Constant Constant Constant Constant Constant Constant Constant Constant Constant Constant Constant Constant Constant Constant Constant Constant Constant Constant Constant Constant Constant Constant Constant Constant Constant Constant Constant Constant Constant Constant Constant Constant Constant Constant Constant Constant Constant Constant Constant Constant Constant Constant Constant Constant Constant Constant Constant Constant Constant Constant Constant Constant Constant Constant Constant Constant Constant Constant Constant Constant Constant Constant Constant Constant Constant Constant Constant Constant Constant Constant Constant Constant Constant Constant Constant Constant Constant Constant Constant Constant Constant Constant Constant Constant Constant Constant Constant Constant Constant Constant Constant Constant Constant Constant Constant Constant Constant Constant Constant Constant Constant Constant Constant Constant Constant Con	MyWorks MyWorks MyNoptics H Market My Market My Mynoptics Market My Mynoptics Mynoptics Mynoptics Mynoptics Mynoptics Mynoptics Mynoptics Mynoptics Mynoptics Mynoptics Mynoptics Mynoptics Mynoptics Mynoptics Mynoptics Mynoptics Mynoptics Mynoptics Mynoptics Mynoptics Mynoptics Mynoptics Mynoptics Mynoptics Mynoptics Mynoptics Mynoptics Mynoptics Mynoptics Mynoptics Mynoptics Mynoptics Mynoptics Mynoptics Mynoptics Mynoptics Mynoptics Mynoptics Mynoptics Mynoptics Mynoptics Mynoptics Mynoptics Mynoptics Mynoptics Mynoptics Mynoptics Mynoptics Mynoptics Mynoptics Mynoptics Mynoptics Mynoptics Mynoptics Mynoptics Mynoptics Mynoptics Mynoptics Mynoptics Mynoptics Mynoptics Mynoptics Mynoptics Mynoptics Mynoptics Mynoptics Mynoptics Mynoptics Mynoptics Mynoptics Mynoptics Mynoptics Mynoptics Mynoptics Mynoptics Mynoptics Mynoptics Mynoptics Mynoptics Mynoptics Mynoptics Mynoptics Mynoptics Mynoptics Mynoptics Mynoptics Mynoptics Mynoptics Mynoptics Mynoptics Mynoptics Mynoptics Mynoptics Mynoptics Mynoptics Mynoptics Mynoptics Mynoptics Mynoptics Mynoptics Mynoptics Mynoptics Mynoptics Mynoptics Mynoptics Mynoptics Mynoptics Mynoptics Mynoptics Mynoptics Mynoptics Mynoptics Mynoptics Mynoptics Mynoptics Mynoptics Mynoptics Mynoptics Mynoptics Mynoptics Mynoptics Mynoptics Mynoptics Mynoptics Mynoptics Mynoptics Mynoptics Mynoptics Mynoptics Mynoptics Mynoptics Mynoptics Mynoptics Mynoptics Mynoptics Mynoptics Mynoptics Mynoptics Mynoptics Mynoptics Mynoptics Mynoptics Mynoptics Mynoptics Mynoptics Mynoptics Mynoptics Mynoptics Mynoptics Mynoptics Mynoptics Mynoptics Mynoptics Mynoptics Mynoptics Mynoptics Mynoptics Mynoptics Mynoptics Mynoptics Mynoptics Mynoptics Mynoptics Mynoptics Mynoptics Mynoptics Mynoptics Mynoptics Mynoptics Mynoptics Mynoptics Mynoptics Mynoptics Mynoptics Mynoptics Mynoptics Mynoptics Mynoptics Mynoptics		
COLS			

6: Integrated Plugin

The layout of the Workspace does not need to be saved. All changes are saved automatically to the Workspace.

IMPORTANT

Be careful: even changes to the predefined Workspaces are saved immediately and without comment. Only Workspaces marked with an (R) are "read only". Changes made to these Worspaces are not saved.

The functions in the context menu can also be accessed in the view menu:

• Add Plugin

Opens a list with all **Plugins** available and not integrated into the application. **Plugins** can be integrated into a frame via drag and drop. It is possible to add more than one **Plugin** into a single frame.

If more than one Plugin share a frame, a certain Plugin can be selected by clicking onto its tab.

• Undock Plugin

Undocks the Plugin. The Plugin opens in an external window. For detailed information on docking and undocking Plugins, see Section 4.1.1.

• Move Plugin

Selecting this item in the menu will open up a small dialog showing graphical representations of the different Frame sets within the application. The selected **Plugin**'s frame is shown in dark gray.

Select the destination of the selected **Plugin** by clicking into one of the light gray representations, and the **Plugin** will be moved to the according Frame



Dialog to Choose the Destination of a Plugin

• Remove Plugin

This will remove the selected **Plugin**. After removing a **Plugin** from the application, it will appear in the list of available **Plugins**.

• Split Frame

Choosing this item will split the currently selected frame into two frames.

• Maximize

Maximizes the currently selected Frame. After maximizing a frame, the menu item will be renamed to *Reset View*...

2.3.5 Managing Workspaces

To manage own or predefined **Workspaces**, choose **View** \ **Workspaces** \ **Manage Workspaces**... from the menu. This will open up the **Manage Workspaces** dialog.

Synoptics (R)	
Procedures (R) Stocedures (R)	Create <u>N</u> ew
Documentation (R)	C <u>h</u> ange Name
MyWorkspace	<u>D</u> elete
Move Up Move Down	

The Manage Workspaces dialog

The following actions can be performed in this dialog:

• Create New

This will create a new empty **Workspace** with the name entered in the input field and put it at the end of the list of **Existing Workspaces**.

• Change Name

Click this button to change the name of an existing **Workspace**. This Button will only be enabled if a existing **Workspace** has been selected before.

• Delete

To delete an existing **Workspace**, select it in the list and press the **Delete** button. If the deleted **Workspace** was the one currently selected, the first one in the list will automatically be selected.

• Move Up / Down

This will move the selected **Workspace** up or down in the list of **Existing Workspaces**. The order in the list is the order in which the **Existing Workspaces** will appear in the **Workspaces** sub menu.

2.3.6 Creating Own Windows

It is possible to create own windows in which **Plugins** can be integrated like in each other frame. To create a new Window, select **View \ Windows \ New Window...** from the menu. This will open a dialog which asks for a name for the new window. This name will be displayed in the window title.



Self Created Empty Window

After one or more Windows have been created, they are listed in the **View \ Windows** menu. The way how **Plugin**s can be integrated into them is similar to integrating **Plugin**s into frames of the main window.

Custom windows are directly connected to the **Workspace** in which they have been created. As a consequence they will be only visible if this **Workspace** is currently selected. Even the **View** \ **Window** menu only contains the windows for the current **Workspace**.

If the user closes a window, the application will display a confirmation dialog. After the user confirmed, the window will be closed and its layout will be lost.

The windows (unless they have been closed) will be available after the application has been restarted.

2.4 Preferences

To change preferences for the Lapap Mk II application choose **Options** \ **Preferences** from the menu. This will open the **Preferences** Dialog.

If the application is closed and reopened or Reset Application is chosen from the menu or the tool bar, some of the Preferences will be set to their default values:

- Audio Notification
 - All check boxes *selected*.
- Search
 - Search results layout = *Google Style*.
- Datapool Buffer
 - Auto-export datapool buffer selected.
 - Auto-export incremental *selected*.
- Tool Tips
 - Activate selected.

2.4.1 Audio Notification

This section provides the possibility to associate sound files with specific events in the Lapap Mk II application. Every time an event occurs, the configured sound file will be played.

2.4.2 Search

Here you can select the Search Result Layout. You can select between Google Style and Table Style.

2.4.3 Datapool Buffer

These options are related to the export functionality of the Datapool Buffer.

• Auto-export datapool buffer

This will enable the automated export of the Datapool buffer.

• Auto-export incremental

This option controls the amount of data which is exported on each auto export. if Auto-export incremental is selected, only the updates which arrived since the last export will be saved. If it is not selected, each export will contain the updates of the last 30 minutes.

2.4.4 Tool Tips

Select or delesect Activate if you want to turn on or off the tool tips in Lapap MK II.

Chapter 3

CLSW Status

This Chapter introduces the **CLSW Status Workspace**. This **Workspace** can be used to access the web front ends of the four Columbus CLSW Switches. It provides access to the initial page of the switches. Access to further pages is controlled by ground configuration.

To open one of the four pages, click on the according **CLSW Status** button. The web front end will be displayed in the area below the buttons.

IMPORTANT

) The integration of the CLSW web interface is not yet smooth. See Appendix B for currently known problems.

3.1 The Tabs of the CLSW Browser

The CLSW browser contains three tabs (displayed as three gray buttons below the **CLSW Status** buttons). These tabs contain the following sub pages:

3.1.1 Overview

This page gives a graphical overview about the utilization of each port.

Big View Directory Ontions Heli																								
CLSW Status (R)	- 8			0.0	Ex do	1.65	-		•														Display Kerresh 😈 13	rebu7 12:54:35
💻 🍕 🜌 🖇 🌡 健 🚾		DMS :	S KI	u												Time 00:0	r 1 0:00 :	• • •	7 Timer 00:00	2 1:00 🕂 🛛		Timer 00:00	3 00 - b III 00:00	4
🦂 CLSW 10 Status	W 11 Status	A CLSW	20 Stat	us	-∳: CLS₩ 2	1 Status	;																	
Overview				Port Cou	inters							Port Sta	tus											
40%								F	Port Utili	ization													Legend	
25%																							S Non-Unicast R	Nots Roc Roc
1 0%																							OPart Connected	
1 2 3	4 5	6	7			10	11	12	13	14	15	16	17	18	19	20	21	22	23	24	25 Ø	26	Port Not Connec ØPort Disabled	ted
Status Alert	Date / Time			Descriptio	n	~	<u> </u>							· ·			· ·				~	~		
Loss of link	Feb 13, 200	7 12:45:25 PI	M L	.ost conne	ection to m	ultiple c	devices	on port:	2.	_				_			_	_		_	_	_		
1																								
																		-		. 1				
Refresh																			Open Ev	ent /	Acknowle	edge Sel	ected Events Delete :	Selected Events
Time Classification	Subsystem	Equipment	Link t	o DMS or	tabliched	Commo	nd and	monitori	ing cana	bility for	thic DWC	ic orgila	bla		Text									
13Feb07 12:53:29 NFO	DMS	PWS 3	LOSS	of data lin	nk to DMS:	No com	mand a	and moni	itorina ca	apability	from this	s PWS ava	ulable.											
13Feb07 10:04:55 INFO	ECLSS	CTCU1	USM_	SW_DMC_	USS_Swop	Instanc	e : Mon	itoring Se	ervice: C	TCU1_C	abin_Ten	np1_DMC	back int	o Soft-Li	mit. Activ	e Set 0. I	Currient \	/alue -	19.88.					
12Feb07 16:15:09 INFO	DMS	PWS 3	Link t	o DMS es	tablished:	Comma	nd and	monitori	ing capa	bility for	this PWS	is availa	ble.											
12Feb07 15:47:15 INFO	DMS	PWS 3	Loss o	of data lir	nk to DMS:	No com	mand a	and moni	itoring ca	apability	from this	s PWS ava	ilable.											

CLSW Status Workspace

Below the chart is a list in which events concering the ports arrive. Using the buttons below the list on the right, these events can be handled.

• Open Event

This will open a popup window displaying details about the currently selected event.

🛛 mozembed-linux-gtk1.2		_ 🗆 ×
File		
◆ Loss of Link on port 2	Feb 13, 2007	12:45:25 PM
Description: The connection to the devices on port 2 has been lost.		
Solution: • If the cable has been removed from the port, reattach the cable. • If the device attached by the cable is not active/on, troubleshoot that device. • If the port has been disabled on the other device, re-enable the port on the other devic	e.	
Other Possibilities: The cable is damaged or severed. If so, replace the cable.		
	Acknowledge Event	Delete Event

- To close the dialog, press **Cancel**.
- To Acknowledge the event, press Acknowledge Event.
- To delete the event, press **Delete**.

• Acknowledge Selected Event

Every event is tagged with a small 🗰 icon. This icon indicates, that the event has not yet been acknowledged.

To Acknowledge selected events, just click the Acknowledge Selected Events Button.

• Delete Event

This will delete the currently selected event(s).

3.1.2 Port Counters

This page displays certain counters for each port in a list.

To refresh the list, press the **Refresh** button. To view the details of a port, select it and press the **Details for Selected Port** button.

🔤 ai	pap 🗟	4											
<u>F</u> ile <u>V</u> ie	w <u>D</u> isplay	Options <u>H</u> elp											
cesa) 🔏 Syno	ptics (R)	- 2		😊 🛊 🏠	🎭 💁 🔍	< < ↓ ⇒ +						
<mark>묘</mark> 🍕	S	\rm 🕼 🕼		S DMS	5 Ku								
ak Cl	A CLSW 10 Status CLSW 11 Status & CLSW 20 Status												
Overvi	iew		Po										
Port	MCast Rx	MCast Tx	BCast Rx	BCast I Tx I	Pkts Rx	Pkts T x	Errors Rx						
1	0	0	0	0 0	0	0	0						
2	15	4114	3282	2 4	4570	7402	0						
1	0	0	0	0 1	0	0	0						
4	8218	20	12	21	12340	11/	0						
6	ő	0103	15	9	159	12,000	0						
ž	ŏ	ŏ	ő	0 1	ŏ	ő	ů.						
8	ŏ	ŏ	ŏ	0 0	ŏ	õ	õ						
9	ŏ	ŏ	ŏ	0 0	õ	õ	°						
10	ō	Ō	0	0 0	0	0	0						
11	0	0	0	0 0	0	0	0						
12	0	0	0	0 0	0	0	0						
13	0	0	0	0 0	0	0	0						
14	0	0	0	0 0	0	0	0						
15	0	0	0	0 0	0	0	0						
16	0	0	0	0 0	0	0	0						
17	0	0	0	0 (0	0	0						
18	0	0	0	0 0	0	0	0						
19	0	0	0	0 0	0	0	0						
20	0	0	0	0 1	0	0	0						
21	0	0	0	0 1	0	0	0						
22	0	0	0	0 0	0	0	0						
24	õ	ň	Ň	ň i	ň	ň	ů.						
25	ŏ	ŏ	ŏ	ŏ i	ŏ	ŏ	õ						
26	ō	0	0	0	0	ō	ō						
Refr	esh Deta	ails for Selected F	Port										
Т	Time	Classification	Subsystem	Equipment									
13Feb07	7 12:53:34	INFO	DMS	PWS 3	Link to DM	S established	: Command and monitoring capability for this PWS is						
13Feb07	7 12:53:29	INFO	DMS	PWS 3	Loss of da	ta link to DMS	5: No command and monitoring capability from this P\						
13Feb07	7 10:04:55	INFO	ECLSS	CTCU1	USM_SW_D	MC_USS_SWO	p_Instance : Monitoring Service: CTCU1_Cabin_Temp1						
12Feb07	7 16:15:09	INFO	DMS	PWS 3	Link to DM	S established	: Command and monitoring capability for this PWS is						
12Feb07	7 15:47:15	INFO	DMS	PWS 3	Loss of da	ta link to DMS	5: No command and monitoring capability from this P\						

CLSW Port Counters

- To refresh the list, press the **Refresh** button.

- To view the details of a port, select it and press the **Details for Selected Port** button. This will open the Details page:

🔳 lanan 🧶				
File View Display Ontions Help	1			
The Mew Disblay Obuous Help				
Synoptics (R)	8 680	🛎 † 🏠 🗞 😤 🔍 🤍 🥙 📘		
💻 🍕 🜌 🖇 🌡 🕼 <mark>sw</mark>	S DMS	Ки		Timer 1
A CLSW 10 Status	W 11 Status 🛛 🔺 CLSW 2	0 Status 🛛 🔺 CLSW 2.1 Status		
Overview		Port Counters	Port Status	
			Status and Counters - Port Counters	- 5
Name :				
Link Status :	Up			
Bytes Rx :	12.706		Bytes Tx :	23,582,756
Unicast Rx :	166		Unicast Tx :	6124
Bcast/Mcast Rx :	13		Bcast/Mcast Tx :	12,084
FCS Rx :	0		Drops Tx :	0
Alignment Rx :	0		Collisions Tx :	0
Runts Rx :	0		Late Colin Tx :	0
Giants Rx :	0		Excessive Colln :	0
Total Rx Errors :	0		Deferred Tx :	2
Return to Summary				
Time Classification	Subsystem Equipment			Text
13Feb07 12:53:34 INFO	DMS PWS 3	Link to DMS established: Command	and monitoring capability for this PWS is available.	
13Feb07 12:53:29 INFO	DMS PWS 3	Loss of data link to DMS: No comma	nd and monitoring capability from this PWS available.	
13Feb07 10:04:55 INFO	ECLSS CTCU1	USM_SW_DMC_USS_Swop_Instance :	Monitoring Service: CTCU1_Cabin_Temp1_DMC back	into Soft-Limit. Active Set 0. Current Value
12Feb07 16:15:09 INFO	DMS PWS 3	Link to DMS established: Command	and monitoring capability for this PWS is available.	
12Feb07 15:47:15 INFO	DMS PWS 3	Loss of data link to DMS: No comma	nd and monitoring capability from this PWS available.	

The details page gives a detailed overview about the port. To get back to the previous page, press the **Return to Summary** button.

3.1.3 Port Status

This page displays the current States of each port.

This page offers only one functionality: **Refresh**.

the second second				5									
Mki a	pap 🏼	4											
<u>F</u> ile ⊻i	ew <u>D</u> isplay	Options <u>H</u> elp											
Cesa	a) 💰 Syno	ptics (R)	- 2	680	😂 🛊 🏠 🎙	<u>।</u> २ २ २	ି 🕹 🔶	•	• •				
<mark></mark> 4	1 🖉 🖉	🜡 健 sw		S DMS S	5 Ku								
40	A CLSW 10 Status A CLSW 11 Status A CLSW 20 Status												
Overv	/iew				Port Co	ounters					Р	ort Statu:	5
Port	Port Typ	e Enabled	Link Status	Current Mode	Flow Ctrl	Bcast Limit	:						
1	10/100	TX Yes	Down	10HDx	off	0							
2	10/100	TX Yes	lln	10HDx	off	ó							
3	10/100	TX Yes	Down	10HDx	off	Ó							
4	10/100	TX Yes	dD	10HDx	off	Ó							
5	10/100	TX Yes	dμ	10HDx	off	Ó							
6	10/100	TX Yes	Down	10HDx	off	Ó							
7	10/100	TX Yes	Down	10HDx	off	Ó							
8	10/100	TX Yes	Down	10HDx	off	0							
9	10/100	TX Yes	Down	10HDx	off	0							
10	10/100	TX Yes	Down	10HDx	off	0							
11	10/100	TX Yes	Down	10HDx	off	0							
12	10/100	TX Yes	Down	10HDx	off	0							
13	10/100	TX Yes	Down	10HDx	off	0							
14	10/100	TX Yes	Down	10HDx	off	0							
15	10/100	TX Yes	Down	10HDx	off	0							
16	10/100	TX Yes	Down	10HDx	off	0							
17	10/100	TX Yes	Down	10HDx	off	0							
18	10/100	TX Yes	Down	10HDx	off	0							
19	10/100	TX Yes	Down	10HDx	off	0							
20	10/100	TX Yes	Down	10FDx	off	0							
21	10/100	TX Yes	Down	10FDx	off	0							
22	10/100	TX Yes	Down	10FDx	off	0							
23	10/100	TX Yes	Down	10FDx	off	0							
24	10/100	TX Yes	Down	10FDx	off	0							
25		No	Down		off	0							
26		No	Down		off	0							
	Refresh												
					1								
	Time	Classification	Subsystem	Equipment									
13Feb0	7 12:53:34	INFO	DMS	PW2.3	Link to DMS	established: Co	ommand and	d monito	ring capat	ollity for this	s PWS i:	s available	t
13Feb0	7 12:53:29	INFO	DMS	PWS 3	Loss of data	link to DMS: N	o command	and mo	nitoring ca	pability fro	m this I	PWS availa	ιble.
13Feb0	7 10:04:55	INFO	ECLSS	CTCU1	USM_SW_DM	C_USS_Swop_In	istance : Mor	nitoring S	Service: CT	CU1_Cabir	n_Temp	1_DMC b	ack into Sc
12Feb0	7 16:15:09	INFO	DMS	PWS 3	Link to DMS	established: Co	ommand and	d monito	ring capat	oility for this	s PWS i:	s available	1.
12Feb0	7 15:47:15	INFO	DMS	PWS 3	Loss of data	link to DMS: N	o command	and mo	nitoring ca	pability fro	m this I	PWS availa	ıble.

CLSW Port Counters

Chapter 4

Synoptic Display System

The Synoptic Display System provides online access to the nominal part of columbus' subsystems. Data is directly derived from the DMS data pool and command are initiated directly to the on-board execution software by the crew.

The synoptics show subsystems and equipment in graphical schematics with data representation in alphanumerical output boxes, states, animated symbols and line colorcoding. Commands are predefined so that the crew can issue the command by either clicking on a button or selecting the command from a list then clicking **Execute**.

The displays in Lapap Mk II are organized in a hierarchical way. There is a main display showing an abstract overview of the whole columbus module with all its subsystems. It is by default configured as the **Home Display** and will be loaded automatically at application startup. It can be opened by clicking onto the **Load Homepage** button in the toolbar or in the **View** menu. From this display it is possible to navigate to the displays of the different subsystems.

The Synoptic Displays provide commanding for all subsystems, display **Measurement**s acquired by the DMS and indicators of limit supervision and data validity.



The Synoptics Home Page

4.1 Window Handling

This section describes the window handling according to the **Synoptic Displays**. All the functionality described here refers to the **Synoptic Display** area.

4.1.1 Docking / Undocking Displays

Undocking a display means to open the Display in a separate frame which is independent from the **Synoptic Display Executor** frame. This Frame has an own menu bar with the according Menu items.

To undock a Display, select it and choose **Display** \ **Undock** from the menu bar. Alternatively you can perform a right click on the tab of the Display you want do undock and select **Undock** from the context menu.



Lapap Mk II With an Undocked Display

The Menu of the undocked Display contains the structure described below. To find Instructions for the Menu Items follow the links to the referring sections in this manual.

- Window
 - Dock --> Section 4.1.1
 - Close --> Section 2.1.3.1
- Display
 - Find Text As You Type --> Section 2.1.3.10
 - Find Parameter In Other Displays --> Section 2.1.3.11
 - Properties --> Section 2.1.3.12
- Help

- Current Display --> Section 2.1.6

To dock the Display, choose **Window \ Dock** from the Displays menu bar.

4.1.2 Toggling Tabbed Mode

Displays can be shown either on a tabbed window (one window at a time) or as iconized internal windows (many in parallel). This option toggles between these two modes.

In this manual the mode showing the displays as independent internal windows will be called "windowed mode", and the mode showing one window at a time will be called "tabbed mode".

To toggle between the two modes press the icon in the tool bar or select **Display** \ **Toggle Tabbed Mode** from the menu bar.

If the application is set to tabbed mode, each display will be shown in the Synoptic display hierarchy in full size. To switch between displays the tabs have to be used.

While in windowed mode, each display will be shown in a own internal Window which can be resized and moved.

4.1.3 Display Size

4.1.3.1 Changing Display Size

Displays can be maximized within the Synoptic Display area using the maximize button of the internal frame. If a Display is not maximized, it can be resized like any other window by clicking and dragging its borders to the desired size.

4.1.3.2 Synoptic Default Size

Although the size of a Display can be chosen manually, there is a default size for each Display. This original size is the size it has been designed for and in which it can be viewed best.

To restore this default size select **Display** \ **Synoptic Default Size** from the menu or click the referring icon in the tool bar.

IMPORTANT

This does only work if the synoptic display area is not set to Tabbed Mode.

4.2 Synoptic Navigation Buttons

Every Display contains specific navigation buttons on the right side. These buttons represent the different subsystems of the columbus module and selecting one of them opens the "Functional Overview" - display of the according subsystem. The home display contains an additional Button to close itself. Every sub display contains a button at which (re)opens the home display.



The Synoptic Navigation Buttons

For every open display a tab is displayed on top of the "synoptic display area" allowing the crew to return to an previously opened display by clicking on it.

4.3 **Display Features**

4.3.1 Overall Monitoring Status

The overall **Monitoring Status** of **Measurements** on the particular display is indicated by the small indicator in the title (or tab) of a display. If one of the **Measurements** is out of its soft or hard limits, it turns light orange or orange. It always has the color of the **Measurement** with the most critical **Monitoring Status**. (I.e. the overall **Monitoring Status** of the **Home Display** shown here is shown as an orange dot, because some **Measurements** within the display are out of their hard limits).



Overall Monitoring State Indicator

4.3.2 Element Types

Synoptic - Displays can contain different types of command or control elements which offer different functionality.

4.3.2.1 Navigation Buttons

Rectangular buttons with sharp edges are Buttons which will open displays of certain sub components.

4.3.2.2 Command Buttons / 2 Step Commanding

This special kind of button can be identified by its round edges. These buttons have a different kind of behaviour as the normal buttons within Lapap Mk II. They do not open displays or have other navigation features. These Buttons are to execute commands within the subsystems.

Instead of immediately executing a command when pressing it, the button will remain pressed until the **Execute** Button within the display is clicked. This **2 Step Commanding** is a way to help the user avoid making mistakes while executing commands, as he has to select the command an execute it in two different steps rather than simply clicking onto a button.

4.3.2.3 Measurements

Measurements are numerical representations of certain values in different subsystems (i.e. temperature, water flow, pressure, etc.). They are usually displayed as output boxes with black text and a white background. They change their background color according to certain situations. These colors indicate their **Monitoring Status**. The following table illustrates the meaning of each possible background color.

Table 4.1: Meaning	of Measurements	Background Color
Table 1.1. Interning	or medio di cinterno	Ducing Counter Coror

Co	lor	Valid	Monitored	Out of Soft Limits	Out of Hard Limits
	White	Х			
	Pale Green	Х	Х		
	Light	x	x	x	
Or	ange	Х	Х	Χ	
	Orange	Х	X		Х
	Purple				

• White

The value is valid and within its normal range.

• Pale Green

The value is valid, within its normal range and monitored.

• Light Orange

The value is beyond its soft limits.

• Orange

This value is beyond its hard limits.

• Purple

The value is not valid. This situation may occur if the connection to the DMS is lost. Additional a status indicator "M" is shown. (M means "missing).

There are two important properties for each measure: **Monitoring Status** and **Acquisition Status**. These Properties can be viewed in detail in the **Properties** Dialog of a certain **Measurement**.

• Monitoring Status

Monitoring means that a **Measurement** is observed by a mechanism determining if the value is within the ranges of its soft and hard limits, or is as expected.

• Acquisition Status

The acquisition of a **Measurement** value is the point of time at which the value was retrieved from the data pool.

4.3.2.4 Graphical Elements

Within certain displays, there are a number of inactive elements (like arrows, lines and symbols) which do not provide any functionality except changing its color or appearance depending on the values and states of certain **Measurement**s.

4.3.3 Display Context Menus

Every display contains elements that provide a context menu. It can be opened by clicking the right mouse button within the display. Depending on the display area location where the context menu is instantiated different options are provided:

If the context menu is opened by clicking into an empty area within the display, these general options will appear:
• Copy Display Title to Clipboard

Copies the exact title of the current display into the systems clipboard. This functionality is useful if the user wants to use the exact title of the display in another application (i.e. Open Office).

• Show Information...

Opens help for the current display within the Documentation Plugin.

These general options are also available in the context menus of all other element types within a display.

4.3.3.1 Context Menu for Buttons

Buttons provide a special properties dialog within the context menu. The items in this context menu are:

• Properties

The Properties Dialog for the button contains two tabs with the following properties:

- Tab: General
 - * Enabled: values {yes | no}, if yes, the button will work.
- Tab: On Press
 - * Action: The action this button will perform. In General it will open the display of a component within the subsystem.
 - * New Window: values {yes | no}, if yes, the display linked to this button will be opened in a new window (or tab)

4.3.3.2 Context Menu for Labels

Labels provide the function **Copy Label to Clipboard**. Choosing this menu item will copy the caption of the label into the clipboard.

4.3.3.3 Context Menu for Measurements

Measurements offer the most complex context menu of all element types. The context menu will look slightly different depending on what type the current **Measurement** is. As the differences are only slight, this section does not deviate between the both types.

PROPERTIES

• Properties

The Properties Dialog provides detailed information about a **Measurement** or a **Computation**.

IMPORTANT

The values in the **Properties** dialog of a **Measurement** are not updated by default. That means the displayed values are retrieved when the dialog is opened and not further updated and may deviate from the ones which can be seen in the **Synoptic Display**. Select the checkbox **update** in the top of the dialog to enable real time updates.

The following list contains the most important properties:

- Data Source Type

values {Telemetry | Computation}.

* Telemetry

A Telemetry is a value which is directly measured within a component (i.e. a temperature, water flow, etc.)

* Computation

A Computation isn't a directly measured value. It's a value which is the result of a calculation that combines different values.

– Opsname

A string containing a short, unique name for the **Measurement**.

– Pathname

The Pathname is the hierarchical position of the **Measurement** in the MDB (Mission Data Base).

– SID

The SID is another unique identifier for the currently selected **Measurement**. In this case a number.

Formatted Value

This value represents the **Measurement** as it will be displayed in the **Synoptic Display**. The source of this formatted (and displayed) value depends on the property Output.

– Raw Value

The raw value is the raw data derived from the source of the **Measurement**. I.e. a temperature sensor provides the measured temperature not "as is", but as a value that has to be interpreted.

- Engineering Value

The engineering value is a value calculated from the raw value of a **Measurement**. I.e. the temperature measured by a temperature sensor calculated from its raw data.

Engineering Unit

The unit in which the **Measurement** is measured. For example: A temperature measured in $^{\circ}$ **C** will have the engineering unit **degC**.

– Type

The kind of data. (i.e. a numeric value, a state like ON or OFF, etc.)

- Output

This property controls the source for the formatted value. Values can be "Engineering value" or "Raw value".

- Acquisition Status

Indicates whether the value is acquired (values are reported from the data source) or not (no connection to the DMS data pool could be established or the data could not be acquired by the DMS)

- Monitoring Status

Indicates whether the value is monitored or not.

– Soft Limit (lower) / Soft Limit (upper)

The borders for the soft limit of this value.

- Hard Limit (lower) / Hard Limit (upper)

The hard limit borders of this value.

IMPORTANT

The initial limits for each **Measurement** are configured within the MDB (Mission Data Base). For information related to limits see Section 4.4.

- Expected Value

The expected value of the Measurements monitoring. See Section 4.4.2 for more information.

Acquisition Time

The time the current value of this Measurement has been retrieved

- Time Since Acquisition

The time that has passed since the acquisition time has been retrieved. Its the "age" of the value.

• Show In Quick Graph

This menu item opens the Line Graph for the current **Measurement**. See Section 4.3.4 for more information.

• Find Parameter In Other Displays

This functionality is useful if the user wants to find a certain **Measurement** which is referenced by more than one display.

To search for a **Measurement** in other displays move the mouse pointer over the menu item. A sub menu showing the Opsname of the **Measurement** opens. Select the Opsname of the **Measurement** to open the **Find Parameter** dialog.

While searching for displays containing the **Measurement**, a progress bar will be displayed, showing the search progress.

There are some actions available (even while the search is still running):

- Stop

Stops the search. This function is useful if the searched display has already appeared in the search results and no further searching is needed.

- Select All

Selects all displays found.

– Open

Opens the selected displays.

- Close

Closes the "Find Parameter" - dialog

🔤 Find Parameter	
Searching for parameter	
CTCU1_Cabin_Temp1_DMC	
20%	
Displays containing parameter:	
COL Synoptics Home Page (COL_HOME)	Stop
	Select All
	<u>O</u> pen
n	Close

Find Parameter Dialog

• Copy Computation to Clipboard

If the current **Measurement** is a **Computation**, it is possible to copy the rule which is used to calculate it into the systems clipboard.

• Copy Opsname to Clipboard

If the current **Measurement** is a **Telemetry**, it is possible to copy its **Opsname** into the systems clipboard.

• Change Monitoring Values

This will open the **Limit Manager** for this value. The **Limit Manager** may be used to change the soft limits for a **Measurement**. See Section 4.4 for Details.

4.3.4 Quick Graphs

4.3.4.1 Purpose of Quick Graphs

A Quick Graph is a graphical representation of the according **Measurement**. It shows the values of the **Measurement** in relation to the time as a graph. The soft and hard limits are also displayed. Because of this it is easily possible to identify if a value is leaving its limits repeatedly.

Graphs display the value of a certain **Measurement** and its changes over a period of time as a graphical diagram. A Graph can be opened by right clicking upon the value which should be displayed and choosing **Show In Quick Graph** from the context menu.

IMPORTANT

There is always only one Quick Graph open. If "Show As Quick Graph" is selected for another Measurement, The already opened Quick Graph will show both the graphs.

The window of a Quick Graph can be rescaled and the graph will fit its size according to the window. It is also possible to interact with the graph's axes to get a better view of a certain region in the graph.



Line Graph of 4 Numerical Measurements

4.3.4.2 Axis Functionality

The Quick Graph can be scaled and moved along its axis to get a better view of a certain region within the graph.

• Moving Along an Axis

To shift the content of the Quick Graph along an axis, simply click and drag on the axis description. After releasing the mouse, the graph will be redrawn.

• Scaling Along an Axis

To scale the axis, click and drag on the axis description while pressing the CTRL key.

IMPORTANT

To make it easier to find the correct position in the Quick Graph to use one of the functions described above, the cursor will change to a hand cursor.

4.3.4.3 Context Menu of a Quick Graph

A Line Graph offers a context menu with the following entries:

• Save Snapshot

This function saves the current state of the line Graph into a \star . PNG image into the users home directory.

• Show Information

This menu item opens the help dialog of the according display.

• Copy Display Title to Clipboard

This will copy the title of the display according to the **Measurement** into the operating systems clipboard.

The Quick Graph offers a different context menu for its axes. (To open this context menu, move the mouse over an axis description until the cursor changes to a hand cursor):

• Axis Reset

If the axis values have been changed before, this menu item restores the default values for the axis.

• Axis Auto-Range

If this checkbox is selected, the scaling of the axis will be automatically adapted to the values in the graph so that the minimum and maximum value of the axis are within the visible region of the graph.

• Axis Auto-Move

If this checkbox is selected, the graph will scroll to keep the last update within the visible range.

4.3.4.4 Linegraph Menu

If the Line Graph is in an undocked Window, it provides a menu bar with the following structure:

• File

- Create HTML Report

This functionality is useful to generate a detailed report about the current state of the visualized **Measurement**. The Report will be saved into the user's configuration directory

- Save Snapshot

This function saves the current state of the line Graph into a \star .PNG image into the user's configuration directory.

– Print

This will open a common printer dialog to configure the printer settings and to print the current state of the line graph.

Print Preview

To setup the page before printing the line graph, choose this menu point. To take a look at the preview of the page as it would be printed, click the OK button.

- Properties

Choose this menu item to open a dialog showing the properties of the line graph.

– Close

This will close the line graph window.

- Window
 - Dock

Choose this menu point if you want to dock the line Graph window into the Synoptic Display area.

Close

Choose this menu point if you want to Close the Quick Graph.

- Help
 - Current Display

This will open the display help for the display according to the line graph.

4.3.5 Performance Indicators

The Performance indicators in lapap inform inform about the synoptic system's performance and the source of the displayed time in the toolbar.



Performance Indicators

• The first indicator item is the **Display Refresh Indicator**. It informs about the update latency of the display. Update latency is the time between receiving the data and finishing the rendering of the display. It can be found at the left of the applicatons time in the toolbar.

	Display Refresh: 🧿	
Performance Indicator	Shows: Up	odate La

tency < 800 ms

If the update latency is less than 800 milliseconds the icon is green.

Display Refresh: 🥥

Performance Indicator Shows: Update Latency between 800 and 2000 ms

If the latency is between 800 and 2000 milliseconds the color changes to light orange.

Display Refresh: 😐

Performance Indicator Shows: Update Latency > 2000 ms, data is dropped

If its color is orange the latency is bigger that 2000 ms. That means that data is dropped, because data from interface cannot be handled.

• The second item concerns the time distribution. It indicates whether the time is derived from the DMS or not.



Time Field: DMS Derived Time

If the time displayed in the toolbar is derived from the DMS, it will be displayed with a gray background.

Di	colov Pofro	ch 🔴	CPS 14Iu	n07 10-1	3.52
- L	эргау кепте	511 0		107 10.1	
3 Time	er 3		Timer 4		_ @

Time Field: Local Time

If the DMS time can't be derived, the local time will be displayed on a light orange background.

More information about the communication state of the DMS can be found in Chapter 5.

4.4 Limit Manager

The **Limit Manager** is a tool to change the monitoring values of a **Measurement**. It is realized with **Synoptic Displays**. Depending on the type of the **Measurement** (numerical **Measurements** or State codes based on discreet **Measurements**) the according display will be opened allowing the user to set the soft limits or to declare an expected value for the **Measurement**. As the two types of displays are different, they are described separately.

To open the **Limit Manager** select **Change Monitoring Values** from the context menu of a **Measurement**.



Opening The Limit Manager For a Measurement

4.4.1 Changing Monitoring Values For Numerical Measurements

● COL Synoptics H ● Quick Graph ● ECLSS Functiona ●	EPDS Functional			
			Current Value	22.0
Enable Monitoring			Disable	Monitoring
DMC_Sys_Monitoring_Table_SW			DMC_Sys_Monitorir	ng_Table_SW
Set Limits	Selected Limits		Current Limits	
	Hard High:	30.689655	Hard High:	30.689655
Soft High: 🔊	Soft High:	30.689655	Soft High:	30.689655
Soft Low: 🔊	Soft Low:	16.0	Soft Low:	16.0
	Hard Low:	16.0	Hard Low:	16.0
	Hard nCount:	30	Hard nCount:	30
Soft nCount: 🔊	Soft nCount:	30	Soft nCount:	30
	Set Lim	its		
	1		1	Close

Limit Manager (Numerical Measurement) The display consists of the following elements:

• Measurement Name (in the display title) and current value

as the **Limit Manager** is realised as a synoptic displays, the value displayed on the right offers the same context menu as other **Measurement**s.

• Command Buttons to enable / disable monitoring

Using this command buttons enables / disables the monitoring of the current **Measurement**. As usual in **2 Step Commanding**, the command will only be executed if the **Execute** button on the bottom of the display is clicked.

• Current Limits section

Displays the current limits of the Measurement. The fields are read only and only for verification.

Set Limits section

New Limits can be entered into the input fields. Besides the values are command buttons to set the values with **2 Step Commanding.** See Section 4.3.2.2.

To change a limit, perform the following steps:

1. Click onto the value that should be changed

The displayed value turns into an editable input box

- 2. Enter the new Value
- 3. Click the command button besides the changed value

This will enable the **Execute** button at the bottom of the display.

4. Click the Execute Button at the bottom of the display

The new value will be updated immediately and can be verified in the current limits section after the next data pool update.

If invalid data or a value that is not allowed has been entered, a message window will open and inform the user. The invalid value will be replaced with the current setting.

Changeable values:

• Soft High Limit / Soft Low Limit

The upper / lower border of the soft limit range. These values may only be changed to values within the hard limits of the **Measurement**. The **Soft High Limit** has to be greater than the current **Soft Low Limit** and the **Soft Low Limit** has to be smaller than the current **Soft High Limit**.

• Soft nCount

This value determines how often the monitored **Measurement** may leave the range of its soft limits before a SAW Message will be generated.

4.4.2 Changing Monitoring Values For State codes (Based on Discreet Measurements)

COL Synoptics H ↓ ● Quick Graph ↓ ● ECLSS Functiona ↓ ●	EPDS Functional	TCU 🔴 ISPR 02 (PL_ISP 🍯	DMS Functional 📔 \varTheta Monitoring (EMU1
			Current Value	READY
Enable Monitoring			Disa	ble Monitoring)
DMC_DMS_Monitoring_Table_SW			DMC_DMS_Monito	oring_Table_SW
	Selected Limit	s	Current Limits	
	Exp. Value:	READY	Exp. Value:	READY
	nCount:	1	nCount:	1
	I			Close

Limit Manager (Discreet Measurement)

The display for discreet **Measurements** consists of basically the same elements as a display for numerical ones, except:

• Set Limits section

The **Expected Value** can be selected from the drop down box rather than typing the new value. The **nCount** value can be entered into the input field. Besides the values are command buttons to set the values with **2 Step Commanding**.

Changeable values:

• Expected Value

As discreet **Measurements** can only be set to specific values, there is no "range" in which the value should be. This is why discreet **Measurements** have no soft or hard limits. Instead they are checked for an expected value.

Soft nCount

This value determines how often the monitored **Measurement** changes into a State code which is different from the expected value.

Chapter 5

Status Displays

There are four icons in the Lapap tool bar which inform the user about the communication state of certain elements. These elements are displayed using different colors to inform about the state of the system they represent.

Depending on which element is considered, these colors have to be interpreted differently.



Status Displays

5.1 Station Mode

As long as values for the Station Mode can be acquired, its icon will indicate the current value of the Station Mode. Its background color depends on the current mode.

If the Station Mode value could not be acquired, the icon will be displayed with a purple background, showing a question mark.

Symbol	Meaning
	Standard
672	Microgravity
	Reboost
	Assured Safe Crew Return
	Proximity Operations
	External Operations
6	Survival
?	Value can't be acquired.

Table 5.1: Station Mode values

5.2 Station Mode Compatibility

The Station Mode Compatibility indicator shows if the current Columbus mode is compatible to the ISS station mode:

- **pale green**: Status measurement could be acquired, Columbus mode is compatible with the current ISS station mode.
- **orange**: Status measurement could be acquired but the current Columbus mode is not compatible with the ISS station mode.
- **purple**: Status measurement could not be acquired.

5.3 DMS Status

The status of the DMS is indicated using two different visual components. The first one is the DMS Status Display and the second one is the DMS time shown in the Date and Time Group.

Depending on the acquisition state of three certain **Measurements** in the data pool the background color of the DMS Status Display is chosen.

- **pale green**: One or more **Measurements** can be acquired and a physical connection to the DMS therefore exists.
- gray: None of the defined **Measurements** can be acquired and a loss of connection to the DMS is assumed.

The background color of the DMS time in the Date and Time Group depends on its acquisition state:

- gray: DMS time can be acquired.
- light orange: Loss of DMS time distribution.

See Section 4.3.5 for more information about the DMS time.

5.4 S-Band / KU-Band

The link state indicators use the following policy to determine their background color:

- **pale green**: Status measurement could be acquired, a connection is present.
- gray: Status measurement could be acquired but its value indicates that no connection is present (Loss Of Signals).
- **purple**: Status measurement could not be acquired. No connection is present.

Chapter 6

System Message Panel

This chapter describes the usage of the **System Message Panel (SMP)**. The **SMP** is used to display the messages which are send by the different subsystems.

The SMP is divided into two different views:

- The **System Message Panel** at the bottom of the Lapap application window displaying the last 5 messages. This view cannot be filtered or sorted.

- A view called **Message Buffer View** which allows the user to view all messages. It allows the user to filter and sort the messages and to save and load a set of messages

Another set of components related to this functionality can be found in the tool bar. The SAW Icons.



The SAW Icons

The **Message Buffer View** in combination with this icons is needed to handle the Situational Awareness - functionality. See Chapter 7 for more information.

6.1 The Message Buffer View

The **Message Buffer View** is a separate window displaying all or a subset of the messages. There are several ways to open it:

- By right-clicking onto the header of the **System Message Panel** at the bottom of the Lapap application window and selecting **Show Message Buffer** from the context menu.
- By double-clicking onto the header of the System Message Panel.
- By choosing **Show Message Buffer...** in the **View** menu.
- By clicking The Message Buffer View Icon in the Toolbar. See Section 2.2
- By clicking onto a SAW Icon in the tool bar.

Some Messages will be displayed in a different color (i.e. OUT OF LIMIT) to show their **Situational Awareness** context.





To get a better view of the text of a certain message without resizing the **Message Buffer View** just double - click the message. A tool tip will be displayed showing the full text.

SW	LAPAP	LAPAP_SW_Swop_Instance_Laptop3_MCD_Executor : DMS_MONIT_DIs_Monitoring_Cmd_SW (MT_ID: DMC_Sys_Monitoring_Table_SW, EI_ID: CTCU1_Cabin_Temp1
ECLSS	CFA1	CFA1_Activation_AP : Called FLAP Returned with Status Halt Failure
DMS	DMC	CFA1_Activation_AP : Monitor Enable Cmd in MT_Table DMC_Sys_Monitoring_Table_SW for End_Item CFA1_Input_Current_DMC Failed with Return Status ALR_MONI
DMS	DMC	CFA1_Activation_AP : Monitor Enable Cmd in MT_Table DMC_Sys_Monitoring_Table_SW for End_Item CFA1_Delta_P_DMC Failed with Return Status ALR_MONITORED
DMS	27Feb07 10:38:1	1 OUT OF LIMIT DMS DMC
ECLSS	CFA1_Activation	_AP : Monitor Enable Cmd in MT_Table DMC_Sys_Monitoring_Table_SW for End_Item CFA1_Fan_Speed_DMC Failed with Return Status ALR_MONITORED
ECLSS	CFA1	CFA1_Set_Speed_AP : Called FLAP Returned with Status Success
ECLSS	CFA1	CFA1_Set_Speed_AP : Cmd Execution and Verification Successful
SW	IDF	IDF_Run_SWOP_On_PLCU_AP : Called FLAP Returned with Status Halt Failure
SW	IDF	IDF_Run_SWOP_On_PLCU_AP : SWOP Command Error: LOAD MMC_IDF_SWOP USM_SW_PLCU_MSW_Swop_Instance 0 NOK_NOT_VALID_COMMAND
SW	SCC	SCC_Run_SWOP_On_MMC_AP : Called FLAP Returned with Status Halt Failure
SW	SCC	SCC_Run_SWOP_On_MMC_AP : SWOP Command Error: LOAD MMC_SCC_SWOP USM_SW_MMC_MSW_Swop_Instance 0 NOK_NOT_VALID_COMMAND
ECLSS	CFA1	CFA1 Activation AP : Called FLAP Returned with Status Halt Failure

The tool tip of a message

6.1.1 The columns in the Message Buffer View

Every message in the **Message Buffer View** contains different fields. These are:

6.1.1.1 Time

The on-board time at which the message was generated.

6.1.1.2 Classification

The classification of the message. There are four possible values: "OUT OF LIMIT", "ADVICE", "ERROR" and "INFO". The following list shows the Classifications sorted from important to less important.

• OUT OF LIMIT

"OUT OF LIMIT" - Messages can occur in two possible cases. At first they occur if a monitored measurement leaves its range. The second case in which OUT OF LIMIT messages may appear are critical errors, i.e. if a certain software component in the columbus module crashes.

For more information on measurements and monitoring, see Section 4.3.2.3.

• ADVICE

Indicates a message which should be understood as a message with a higher priority as a message of the type INFO. Though they do not indicate critical errors or malfunctions, users should always check messages of this type.

• ERROR

Indicates the occurrence of an operational or functional error in a subsystem. I.e. the user has tried to execute a command which was not allowed at that point of time.

• INFO

A message with this classification is an general information.

6.1.1.3 Subsystem

The subsystem according to the current message.

6.1.1.4 Equipment

The equipment of the subsystem. A equipment can be understood as a complete functional unit within a certain subsystem, not only a single construction unit (like screws, LEDs, etc).

6.1.1.5 Text

A human readable, textual description of the message.

6.1.2 Sorting the columns of the Message Buffer

When the Message Buffer View is opened, it will be sorted ascending by Time.

The Messages in the **Message Buffer View** (and in the **System Message Panel**) can be sorted by clicking onto the header of the according column. After a column has been sorted, a small arrow in the header of the column shows whether the order is ascending or descending.

It is possible to sort the list according to more than one column. The column which has been clicked most recently provides the main order (i.e "Classification"). The messages are sorted in a way that ensures the entries in the Classification column are ascending. So the previously sorted column ("Sub-system" in this case), is sorted ascending too, but respecting the main order of the Classification column.

To reset the sort, double click on a column. This will apply the main order to it and remove the assortments of all other columns.

🔤 Message	Buffer [I	Live] 🗌		
Situational Awarenes	s: 🗌 DMS 🔲 (EOMMS 🔲 ECL	SS 🔲 EPDS 🛛	TCS Payload SW
Time Filter	ALL 🔻	ALL 🔻	*	*
Time	Classification /	Subsystem 🔺	Equipment	Text
14Jun07 15:28:40	ERROR	DMS	MMC	MMC_Recovery_AP : HW to SW Mapping Variable V
14Jun07 15:26:39	ERROR	DMS	MMC	MMC_Recovery_AP : HW to SW Mapping Variable V
14Jun07 15:26:08	ERROR	DMS	MMC	MMC_Recovery_AP : HW to SW Mapping Variable V
14Jun07 15:22:36	ERROR	DMS	MMC	MMC_Recovery_AP : HW to SW Mapping Variable V
14Jun07 15:21:36	ERROR	DMS	MMC	MMC_Recovery_AP : HW to SW Mapping Variable V
14Jun07 15:18:34	ERROR	DMS	MMC	MMC_Recovery_AP : HW to SW Mapping Variable V
14Jun07 15:16:29	ERROR	DMS	MMC	MMC_Recovery_AP : HW to SW Mapping Variable V
14Jun07 14:50:16	ERROR	DMS	MMC	MMC_Recovery_AP : HW to SW Mapping Variable V
14Jun07 13:34:57	ERROR	DMS	MMC	MMC_Recovery_AP : HW to SW Mapping Variable V
14Jun07 13:09:51	ERROR	DMS	MMC	MMC_Recovery_AP : HW to SW Mapping Variable V
14Jun07 11:54:33	ERROR	DMS	MMC	MMC_Recovery_AP : HW to SW Mapping Variable V
14Jun07 11:04:21	ERROR	DMS	MMC	MMC_Recovery_AP : HW to SW Mapping Variable V
14Jun07 10:39:15	ERROR	DMS	MMC	MMC_Recovery_AP : HW to SW Mapping Variable V
14Jun07 09:49:03	ERROR	DMS	MMC	MMC_Recovery_AP : HW to SW Mapping Variable V
14Jun07 09:23:57	ERROR	DMS	MMC	MMC_Recovery_AP : HW to SW Mapping Variable V
14Jun07 08:33:45	ERROR	DMS	MMC	MMC_Recovery_AP : HW to SW Mapping Variable V
14Jun07 07:18:25	ERROR	DMS	MMC	MMC_Recovery_AP : HW to SW Mapping Variable V
14Jun07 15:24:38	ERROR	ECLSS	CFA1	CFA1_Pwr_On_AP : Cmd CFA1_Pwr_On_DMC Failed
14Jun07 15:22:06	ERROR	ECLSS	CFA1	CFA1_Pwr_On_AP : Cmd CFA1_Pwr_On_DMC Failed
14Jun07 15:19:04	ERROR	ECLSS	CFA1	CFA1_Pwr_On_AP : Cmd CFA1_Pwr_On_DMC Failed
14Jun07 14:37:43	ERROR	ECLSS	CFA1	CFA1_Pwr_On_AP : Cmd CFA1_Pwr_On_DMC Failed
14Jun07 13:22:24	ERROR	ECLSS	CFA1	CFA1_Pwr_On_AP : Cmd CFA1_Pwr_On_DMC Failed
14Jun07 12:07:06	ERROR	ECLSS	CFA1	CFA1_Pwr_On_AP : Cmd CFA1_Pwr_On_DMC Failed
14Jun07 10:51:48	ERROR	ECLSS	CFA1	CFA1_Pwr_On_AP : Cmd CFA1_Pwr_On_DMC Failed
14Jun07 09:36:30	ERROR	ECLSS	CFA1	CFA1_Pwr_On_AP : Cmd CFA1_Pwr_On_DMC Failed
14Jun07 08:21:12	ERROR	ECLSS	CFA1	CFA1_Pwr_On_AP : Cmd CFA1_Pwr_On_DMC Failed
14Jun07 15:27:40	ERROR	EPDS	PDU1	PDU1_SUP1_Pwr_Out1_On_AP : Cmd PDU1_SUP1_
14Jun07 15:25:38	ERROR	EPDS	PDU1	PDU1_SUP1_Pwr_Out1_On_AP : Cmd PDU1_SUP1_
14Jun07 15:24:08	ERROR	EPDS	PDU1	PDU1_SUP1_Pwr_Out1_On_AP : Cmd PDU1_SUP1_
14Jun07 15:19:35	ERROR	EPDS	PDU1	PDU1 SUP1 Pwr_Out1 On_AP : Cmd PDU1 SUP1
14Jun07 14:25:10	ERROR	EPDS	PDU1	PDU1 SUP1 Pwr Out1 On AP : Cmd PDU1 SUP1
14jun07 12:19:39	ERROR	EPDS	PDU1	PDU1_SUP1_Pwr_Out1_On_AP : Cmd PDU1 SUP1
14jun07 11:16:54	ERROR	EPDS	PDU1	PDU1_SUP1_Pwr_Out1_On_AP : Cmd PDU1_SUP1
14lun07 10:14:09	ERROR	EPDS	PDU1	PDU1 SUP1 Pwr Out1 On AP : Cmd PDU1 SUP1
141.07.00.11.74	EDDOD	EDDC	DDU1	PDU1 CUP1 Due Out1 On AB - Card PDU1 CUP1

Sorted by "Classification" as the main order and then by "Subsystem".

6.1.3 Actions in the Message Buffer View

On the right side of the **Message Buffer View** window are different buttons which offer functionality to filter the list, to load and save sets of messages and to freeze and unfreeze the list.

6.1.3.1 Filter

The filter functionality can be used to find a certain message. Also the amount of displayed messages in the *Message Buffer View* can be decreased. There are several criteria according to which the messages can be filtered.

The Situational Awareness check boxes at the top of the Message Buffer View are mentioned in Chapter 7.

Situational Awareness: 🗌 DMS 📄 COMMS 📄 ECLSS 📄 EPDS 📄 TCS 📄 Payload 📄 SW							
Time Filter	N/A	Ŧ	N/A	•	*	*	Reset Filter
Time 🔻	Classificatio	n	Subsyste	m	Equipment	Text	

The Filter Controls for the Message Buffer View

• Filtering according to the message time:

Clicking on the **Time Filter...** Button opens the **System Message Time Filter** dialog. This dialog can be used to define a Filter for the time of the messages.

Mkii Sy	stem Me	ssage	Time Filte	er 🧕		X
Messa	lge Filter					
T:						\odot
Time.	I ALL	<u> </u>	jince 🔾	<u>U</u> ntil	○ <u>R</u> ange	
			Apply	R <u>e</u> set	Cance	:

The System Message Time Filter

The following filtering options can be selected:

- N/A: The messages will not be filtered according to the time they arrived
- Since: All messages since this point in time will be displayed in the Message Buffer View
- Until: All messages that occurred earlier than this point of time will be displayed
- Range: All messages within the selected range will be displayed

If any option except N/A is selected, one or both text fields will be enabled. A click on the clock symbol next to an enabled text field open a simple calendar dialog. This dialog can be used to select a time for one of the text fields. The title of the calendar dialog indicates which date will be configured.

MKI Da	te Afte	er Sele	ection	9					
June 🗧 2007 🗧									
Sun	Mon	Tue	Wed	Thu	Fri	Sat			
28	29	30	31	1	2	3			
4	5	6	7	8	9	10			
11	12	13	14	15	16	17			
18	19	20	21	22	23	24			
25	26	27	28	29	30	1			
2	3	4	5	6	7	8			
2007/165 9 ⁺ / ₊ : 56 ⁺ / ₊ : 21 ⁺ / ₊									
01		Cancel		<u>C</u> lear	N	0W			

Calendar Dialog

In this calendar the desired point of time can be selected. Click the button **OK** to set the selected point of time in the filter dialog, or click **Clear** to remove an existing selection. The button **Now** selects the current time in the calendar dialog.

• Filtering according to the other fields of the message:

There are two drop down and two text fields for the related attributes of a system message. Use these text fields to search for messages with a certain attribute.

Classification

This drop down box can be used to select the classification of the messages which shall be found

- Subsystem

This drop down box can be used to select the subsystem to which the messages shall belong.

- Equipment

Enter search criteria for the field "equipment" into this text field.

– Text

Enter search criteria for the field "text" into this text field.

You may use wild cards in the two text fields. The wild card * (one or more characters) and ? (exactly one character) are supported by the filter.

🔤 Message	Buffer [L	.ive] [F	iltered]				
Situational Awareness: DMS COMMS ECLSS EPDS TCS Payload SW							
Time Filter	OUT OF LI 🔻	SW 🗖	idf	COAP			
Time 🔻	Classification	Subsystem	Equipment				
14Jun07 15:27:09	OUT OF LIMIT	SW	IDF_MMC	COAP_IDF_Swop_Instance_MMC : Inter			
14Jun07 15:23:07	OUT OF LIMIT	SW	IDF_MMC	COAP_IDF_Swop_Instance_MMC : Inter			
14Jun07 15:21:05	OUT OF LIMIT	SW	IDF_MMC	COAP_IDF_Swop_Instance_MMC : Inter			
14Jun07 15:15:56	OUT OF LIMIT	SW	IDF_MMC	COAP_IDF_Swop_Instance_MMC : Inter			
14Jun07 15:15:19	INFO	DMS	LOCALHOST	LAPAP: Link to DMS established: Com			
14Jun07 15:02:49	OUT OF LIMIT	SW	IDF_MMC	COAP_IDF_Swop_Instance_MMC : Inter			
14Jun07 14:12:37	OUT OF LIMIT	SW	IDF_MMC	COAP_IDF_Swop_Instance_MMC : Inter			
14Jun07 12:32:12	OUT OF LIMIT	SW	IDF_MMC	COAP_IDF_Swop_Instance_MMC : Inter			
14Jun07 11:42:00	OUT OF LIMIT	SW	IDF_MMC	COAP_IDF_Swop_Instance_MMC : Inter			
14Jun07 10:01:36	OUT OF LIMIT	SW	IDF_MMC	COAP_IDF_Swop_Instance_MMC : Inter			
14Jun07 07:31:00	OUT OF LIMIT	SW	IDF_MMC	COAP_IDF_Swop_Instance_MMC : Inter			

Filter Filled

IMPORTANT



Messages from the subsystem DMS will always be displayed, even if they don't match the filter criteria.

6.1.3.2 Clear

A click on this button removes every message from the **Message Buffer** list. The messages are simply hidden and an empty list will be displayed. There is no way to permanently delete the messages from the system. To bring back the messages that have been hidden, click **Sync**.

6.1.3.3 Sync

A click on this button "synchronizes" the **Message Buffer View** with the system. This means that all messages since the start of the application will be displayed. All filters will be removed. If old messages had been loaded into the **Message Buffer View**, they will be removed to display the current messages.

6.1.3.4 Freeze

If the **Message Buffer View** is frozen, no new Messages will be displayed in the message list. This behavior is helpful when a user searches for a specific message in the list. So he won't get irritated by new messages appearing in his search results. As long as the Buffer is frozen, the caption of the **Freeze** - button will change to **Live**.

If the messages in the view are saved to a *.CSV - file while it is frozen, only frozen messages will be saved to the file.

6.1.3.5 Live

The "Live" - button unfreezes the **Message Buffer View** and all messages that have been sent while the **Message Buffer View** was frozen will appear in the list. The **Live** - button also removes loaded message snapshots from the buffer and returns to the current messages.

6.1.3.6 Load

It is possible to load a set of old messages into the **Message Buffer** using the **Load** - Button. After a set of old messages has been loaded, only these old messages will be displayed in the **Message Buffer**. To return to the live messages click onto the **Live** button. The loaded messages will then be discarded.

6.1.3.7 Save

To save the messages which are currently in the **Message Buffer View**, use the **Save** - Button.

IMPORTANT

All live (or frozen) messages will be saved to the file, even if they do not match the current filter settings. Instead of saving only matching messages the filter settings are put into the $\star.CSV$ - file. After a reload the filter settings in the file are set automatically !!!

6.1.3.8 Acknowledge & Close

If the **Message Buffer** was opened in situational awareness context this button is enabled. See Section 7.5.1 for more information.

6.1.3.9 Close

This button simply closes the **Message Buffer View**.

6.1.3.10 Showing Event Help for a message

To retrieve further information about a message it is possible to display event help by right clicking onto a message and selecting **Show Message Help** from the context menu. The event help will be opened in the main window of Lapap Mk II, after the **Documentation Worspace** has opened. To get back to the display which was open before select the according workspace in the **Workspace** chooser.

6.1.4 The Icons in the Message Buffer View

Above the **Filter** - button are three icons displaying the current state of the **Message Buffer View**:



The Icons in the message Buffer View

6.1.4.1 The "Live" Icon

The **Live** icon on the left indicates that the messages currently displayed in the **Message Buffer View** are synchronized with the application. In this case a green dot is displayed. If the **Message Buffer** is in frozen state the dot is gray.

6.1.4.2 The "Loaded Messages" icon

the middle icon displaying a floppy disc indicates that the messages in the **Message Buffer View** are loaded from a *.CSV - file. If the **Message Buffer View** is in live - mode the floppy disc is displayed gray.

6.1.4.3 The "Filter" icon

If a filter is applied to the **Message Buffer View** a small funnel icon is displayed. If no filter is applied the funnel is displayed gray.

Chapter 7

Situational Awareness

This chapter describes the Situational Awareness (SAW) functionality of Lapap Mk II.

The Situational Awareness functionality of Lapap Mk II informs the user about certain important events. I.e. if a value runs out of its monitoring limits a message with the classification **OUT OF LIMIT** will appear. To make sure that the user recognizes these situations (i.e. in the case that he has not been in front of the screen all the time) **SAW** provides a way to show him that certain events occurred in his absence.

The **SAW** functionality does not change the systems state in any way. It is only providing functionality to inform the user about occurrences of certain events.

7.1 SAW Messages

Messages which are trigger messages for **Situational Awareness** will be displayed in orange in the **System Message Panel** and in the **Message Buffer View**. It is possible to filter the **Message Buffer View** according to messages with **SAW** context. This can be achieved by using the check boxes at the top of the Message Buffer view.

7.2 SAW Icons

The **SAW** icons can be found in the main tool bar of Lapap Mk II. There is an Icon for every subsystem showing whether there is a occurrence of a **SAW** message for the subsystem or not. If a **SAW** message appears for a subsystem, its icon will be displayed with an orange background. This makes it possible to easily recognize every occurrence of **SAW** messages without searching for **SAW** messages by hand.

Mkii	ара	эр	9					
<u>F</u> ile	⊻iev	v <u>D</u> i	splay	Do	cume	ntati	on	0
C	esa	3	Syno	ptics	(R)			
	a de la comercia de l	Ħ	Ş	l	Þ	sw		
COL Synoptics Home Page								

The SAW Icons in the Tool Bar of Lapap Mk II.

7.3 Opening the Message Buffer View for SAW

Though it is possible to search for **SAW** Messages by hand, lapap offers a work flow to simplify this process.

The **Message Buffer View** can be opened by clicking onto one of the **SAW** icons in the tool bar of Lapap Mk II. In the case that the clicked icon shows an orange background, the **Message Buffer View** will be opened in **SAW** Mode. The button **Acknowledge & Close** near the bottom right corner of the dialog is now enabled.

if a subsystem icon is clicked which shows a gray background, the **Message Buffer View** will be opened without **SAW** context. But a filter setting will be applied to show only messages according to the subsystem of the clicked icon.

To open the Message Buffer view for **SAW** just click on a **SAW** icon which shows a orange background. This indicates that there are **SAW** messages for the according subsystem. The view can also be opened by right - clicking onto the icon and selecting **Show Message Buffer** in the context menu.

🎟 Messag	e Buffe	er - Situ	uation	al Awareness: Subsystem [ECLSS] [Live] [Filtered] ③			
Situational Awareness: DMS COMMS ZELSS EPDS TCS Payload SW							
Time Filter	OUT OF LI 🔻	ECLSS 🚽 🔻	*	*			
Time	Classification A	Subsystem 🔺	Equipment	Text			
01Mar07 11:15:40	INFO	DMS	PWS 3	Link to DMS established: Command and monitoring capability for this PWS is available.			
27Feb07 10:38:13	OUT OF LIMIT	ECLSS	CFA1	CFA1_Activation_AP : Called FLAP Returned with Status Halt Failure			
27Feb07 10:14:25	OUT OF LIMIT	ECLSS	CFA1	CFA1_Activation_AP : Called FLAP Returned with Status Halt Failure			
26Feb07 08:29:56	OUT OF LIMIT	ECLSS	СТСИ	CTCU_Ena_Cabin_Air_Mon_AP : Called FLAP Returned with Status Halt Failure			

The Message Buffer View opened for SAW.

7.4 SAW Filter Functionality

Situational Awareness Messages can be filtered using the check boxes at the top of the Message Buffer View. If one of these check boxes is selected, the dialog will display only messages of the according subsystem defined to be trigger messages for **SAW**.

The Message Buffer View is now in Situational Awareness Mode. The dialog title changes and enumerates the selected subsystems.

It is possible to use the standard filter functionality while the Message Buffer is in Situational Awareness Mode. But the drop down boxes to select the classification and the subsystem are disabled as the **SAW** Mode implies search criteria for these fields.



The Message Buffer View Filtering SAW Messages

7.5 Handling SAW messages

The **Message Buffer View** offers all the functionality it has in its original context. But only **SAW** messages according to the selected subsystems will be displayed.

7.5.1 Acknowledging SAW messages

To Acknowledge the **SAW** message for the according subsystem click the button **Acknowledge & Close**. The **Message Buffer View** will be closed and the icon's background color changes from orange to gray.

Acknowledging a set of **SAW** messages has the following consequences: The **Message Buffer View** will be closed and the according icon will change its background color back to gray. If new **SAW** messages appear for the according subsystem, the messages acknowledged before will not be displayed again if the **Message Buffer View** is opened in **SAW** context for the same subsystem the next time.

It is possible to acknowledge **SAW** messages without opening the **Message Buffer View**. Just right - click onto the icon of the subsystem and choose **Acknowledge**. To acknowledge all subsystems at once choose **Acknowledge All**.



The context menu of a SAW icon

Chapter 8

Timer Section

This chapter describes the timer section in the Date & Time Group within Lapap MK II.

The timer section contains four similar timers which can be configured and used for different purposes.



Timer Section of Lapap Mk II

8.1 Timer Types

The timer section in the tool bar of Lapap Mk II contains four timers. These timers can be configured in two ways:

Alarm Clock Timer

Timer 1 🚫 01:32:54 🕂 🕨 🔲 🚺

These timers can be identified by the alarm clock symbol.

Count Down Timer

Timer 1 3

The stop watch symbol identifies this timer type.

Both timer types count backwards and display the remaining time in an input field. Its background color changes to indicate the remaining time:

- green = remaining time > 1 minute
- **light orange** = remaining time < 1 minute
- **orange** = timer has reached the target time and is counting forwards to show the time that has passed since then.

The difference between the both timer types is the way the time that the user enters into the input field is interpreted.

8.1.1 Alarm Clock Timer

The **Alarm Clock Timer** can be set up to a specified **End Time** (within a 24 hour range). After setting up the **End Time**, a click onto the play button will start the timer. It will display the amount of time that is remaining until the **End Time** is reached. The end time will be calculated according to the time displayed in the upper right corner of the **Date & Time Group**.

This timer type can be used as a reminder which will remind the user about an event which will occur at the specified **End Time**.

To display the **End Time** of the **Alarm Clock Timer**, move the mouse over its label. This will show the **End Time** in a tool tip.

IMPORTANT

The time of the laptop on which the Lapap Mk II application is running is synchronized with the DMS time at application startup. The time displayed in the application window is the time retrieved from the DMS.

Timer 01:3	1 Vince 1 2:13 - Vince 100:
	Alarm Clock Timer
	End Time: 12:00:00

Displaying The Target Time of a Alarm Clock Timer

8.1.2 Count Down Timer

The **Count Down Timer** can be set to a time within the range of 24 hours. After clicking the start button this time will be counted backwards.

8.1.3 Timer Settings

To change the settings of a timer open the context menu by right clicking onto the symbol of the according timer. Select **Settings...** from the context menu.

To reset a timer to its default state, select **Reset** from the context menu which opens on a right click on the timer's label.

Name	Timer 1		
Type Countdown Timer			
Sound ring.wav			
Joana	Apply Concel		

Timer Settings Dialog

- To make the changes work, click **Apply** or press the **ENTER** key.
- To cancel the operation, click **Cancel** or press the **ESC** key.

The following settings apply to both timer types:

8.1.3.1 Name

Each timer has a label that is displayed above the remaining time. To distinguish the different timers it may be helpful to mark them with a custom label.

This label can be reset to its default value by selecting the context menu item **Reset Label.** This can be done while the timer is running.

8.1.3.2 Type

The timer type can be set to Alarm Clock Timer or Count Down Timer.

8.1.3.3 Sound

It is possible to select a sound which will be played when the timer reaches the specified time.

Chapter 9

International Procedure Viewer Components

This chapter describes the components that realise the functionality of the **International Procedure Viewer** (**IPV**) in lapap. Lapap contains components to search, navigate and work through **ODF Procedures** and the related Documentation.

The components are based on the IPV and they work similar.

To execute or work through an **ODF Procedure** means to manually perform the steps described in the procedure.

The IPV functionality contains three major components:

1. **ODF Book Browser**

The **ODF Book Browser** provides a tree view in which all accessible **ODF Procedures** can be found. The root entry of the tree is labeled **ODF Libraries**.

This view can be used to navigate to a specific procedure and to start it. It also provides a way to search for specific procedures.

See Section 9.1

2. Procedure Executor

The **Procedure Executor** is a view in which the procedure selected in the **ODF Book Browser** will be opened. Depending on the type of the selected procedure this view looks different.

This view is used to work with the procedures.

See Section 9.2

3. Documentation Viewer

The Documentation Viewer is useful to search for and display help information on certain topics.

See Section 9.3

9.1 ODF Book Browser

To access the **ODF Book Browser**, select one of the default **Workspace**s **Synoptics** or **Procedures**. Each one of them contains the **ODF Book Browser**.



The ODF Book Browser

9.1.1 Selecting ODF Procedures

The **ODF Book Browser** provides a tree view which contains all available **ODF Procedures**. To select a certain procedure, navigate through the tree to the desired procedure and simply select it. Depending on the workspace that was previously selected, the procedure will open in a different place.

• In Workspace **Synoptics**

The **ODF Book Browser** can be found in the lower left corner of this workspace. By selecting a procedure, the workspace **Synoptics & Proc** will be opened and the procedure will be shown there on the left side while previously opened **Synoptic Displays** remain visible on the right side.

The workspace **Synoptics & Proc** is useful if a specific display is needed to perform the procedure. But as the **ODF Book Browser** is not visible in this workspace, it is necessary to switch to another one to select a different procedure.

• In Workspace **Procedures**

The workspace **Procedures** is focused completely on the work with procedures. It contains the **ODF Book Browser** on the left side and the **Procedure Viewer** on the right side.

9.1.2 Searching For Procedures

The **ODF Book Browser** provides a search functionality to find certain procedures or documentation related to them.

To search for a procedure, enter a search term in the text box in the **ODF Book Browser** and press the **Search** Button.



Searching For Procedures

This opens the **Documentation** workspace, where the search results are displayed.

an lapap 🕘					
<u>File View Display Documentation Options</u> <u>H</u> elp					
Cesa 🔏 Documentation (R) 🔽 🖷 🚍 🚍	* 🚵 🗞 🔊 🔍 🍕 🦓 🖕 • • •				
🖳 🍕 🕖 🖇 🕼 🕼 SW 🕅 🎦 DMS S					
ECLSS Search	Back Forward				
Search: Procedures Documentation Displays All	Crumb Trail: <u>RefDoc</u> > LAPAP Doc Plugin Test				
Search: 🖲 Titles 🔾 Contents 🔾 Both					
1 result(s) found for search : ECLSS					
TABLES	Documentation				
COL ECLSS Parameter Tables Cabin Fan Assembly (CFA) Parameter Table Parameter Explanation Default Default					
/col_data/odf/ECLSS/REFERENCE/5.101_E_ECLSS_5101.xml	This index provides links to information on the following topics:				
	Reference Documentation:				
	 ECLSS Manual EPDS Manual DMS Manual COMMS Manual TCS Manual 				
	◆ Help				
	Crew Event Help Lange Mk Ullser Manual				
	Capap mk ii Osci manual				

Searching For Procedures

Search terms can be specified using the boolean operators **AND**, **OR** and **NOT**. Note that these operators have to be written in uppercase letters to be considered as being operators rather than parts of the query (like in Google).

Examples:

- To Search for a procedure or documentation containing the terms DMS and Overview, use the query "*dms AND overview*".
- To Search for a procedure or documentation which contain only the term DMS but not Overview, use the query "*dms NOT overview*".

If documentation about a certain procedure should be found, the radio button **Documentation** has to be selected and the search has to be restarted. This can be done by pressing the **Search** Button or by typing the **ENTER** key.

It is possible to select the whether the search shall be performed on the titles, the Content or both. This can be done by selecting the according radio button in the **ODF Search Plugin**.

IMPORTANT

If documentation shall be found, it is recommended to search for contents rather than for titles.

9.1.3 Displaying Procedure Properties

To display the properties of a procedure, right-click onto its name. A small dialog will open and display the properties of the selected procedure.

The following properties can be viewed:

• Unique ID

A unique identifier for the procedure.

• Date

The date the procedure was created.

• Procedure code

The Procedure code is a special value which contains information about the procedures purpose and its state of development.

• Format

This property indicates of which format the procedure is. There are three possible procedure formats:

- Logic Flow
- Check List
- PDF Procedure
- Version

A version number.

• Library

The Library in which this procedure is related to.

9.2 Procedure Executor

The **Procedure Executor** displays the content of the selected procedure and provides functionality to perform it. Depending on the type of procedure which is opened it looks different and contains different functionality.

There are currently three types of procedures which can be executed. Each type leads to a different kind of work flow.

9.2.1 Logic Flow Procedures

These type of procedure is designed for work flows which are not linear and contain many decisions. As every decision leads to two or more exits there are many paths which can be followed while the procedure is executed. This type of procedure is best described in a graphical way using symbols for transitions and decisions.

The **Procedure Executor** displays **Logic Flow Procedure**s in a subdivided frame, containing the following two panels:

• Overview Panel

The **Overview Panel** (on the left) provides an overall view of the whole procedure and can be used as a navigation help to keep in mind which part of the procedure is currently displayed in the **Procedure Panel**. The currently displayed section is shown as a green rectangle in the **Overview Panel**.

A slider can be used to change the zoom level of the procedure displayed in the **Procedure Panel**. To enlarge the procedure, move the slider to the right. To get a more distant view, move it to the left. The Button **Return to Default Zoom** sets the zoom level to its default value which is chosen to show as much as possible of the procedure but to keep the procedure readable.

• Procedure Panel

In this panel the procedure is displayed in the selected zoom level. This panel provides functionality to actually work with the procedure currently opened.



A Logic Flow Procedure in the Procedure Executor

9.2.1.1 Working With Logic Flow Procedures

9.2.1.1.1 Elements in Logic Flow Procedures A **Logic Flow Procedure** consists of specific elements which provide different functionalities.

• Instruction Blocks

Each **Logical Flow Procedure** is separated into **Instruction Block**s. There are different types of **Instruction Blocks**:

- Action Blocks

Action Blocks contain a number of procedural steps the user has to perform manually on the equipment or a **Synoptic Display**. An Action Block contains procedural steps which are related to one specific display or equipment.

The single steps can be identified by little black dot in front of them which is called "bullet".



Action Block Containing a Single Procedural Step

- Decision Blocks

Decision Blocks contain one or more procedural steps that require a decision. Therefore a **Decision Block** has two or more exits. Depending on the result of the decision the user has to follow the according exit.



Decision Block With Two Exits

- Combined Action / Decision Blocks

In some cases some procedural steps and a following decision belong together and should be grouped into a single element to make their relationship clear.

The **Combined Action / Decision Block** is such an element combining the two types of **Instruction Blocks**.



Combined Action / Decision Block

Result Blocks

Result Blocks contain a short summary which describes the progress the performing of the procedure has reached at a certain moment of time. No action or decision has to be performed.



• Notes

These small icons provide information about the current step / The whole procedure. To display the information (or to hide it), click onto the icon.



Opening a Note of a Instruction Block

• Step Exits

Step Exits do not just only show to which step they lead, they also contain interactivity. A click onto the description of an exit will select the according next **Instruction Block**.

9.2.1.1.2 Basic Navigation The **Overview Panel** contains a small overview image of the procedure and displays a green rectangle representing the current view port of the **Procedure Panel**. While the scrollbars in the **Procedure Panel** are moved, this rectangle moves simultaneously to show the changed visible region.

Another more direct way to navigate within the procedure is to click into the **Overview Panel** (using any mouse button). This will center the visible region on the position of the mouse click. It is also possible to drag the view to the desired position.

9.2.1.1.3 Executing The Logic Flow Procedures The execution of a **Logic Flow Procedure** has to start with the first available **Instruction Block**, numbered **1**. If additional notes are available, read them first.

To select a certain **Instruction Block** in the procedure, right-click anywhere in the **Procedure Panel**. This will open a context menu displaying the available **Instruction Block**s.





Procedure Panel - Context Menu

After selecting an **Instruction Block** from the context menu, the procedure will be scrolled to make it visible. The selected Block will change its color to light green.

After performing the manual tasks, the next Block can be selected. There are two ways to do this:

1. Clicking onto the description of the exit

This way is the most save way, because the step which has to follow according to the work flow of the procedure will be selected.

2. Selecting the following step from the context menu.

Using the context menu to select the following step is also possible but should be used carefully as it is possible to select a wrong step accidentally.

IMPORTANT

To ensure a proper work flow of the procedure it is recommended to use method 1. rather than the context menu.

9.2.2 Check List Procedures

Check lists are procedures which have a linear flow. The actions are lined up followed one by another and there are no (or very few) decisions which lead to a different path the procedure is walked through. This type of procedure is described best with a checklist.

🛲 lapap 🧕		х
<u>File View Display Documentation Options</u>	Help	
Cesa 🔏 Procedures (R) 🔻	📲 🖷 🚍 🛊 🏠 🐁 🕾 🔍 🧠 🧛 🛶 🔹 🔿 🔹 Display Refresh 🛛 GPS 14Jun07 09:43:-	49
🖳 🍕 🕖 🖇 🔋 🕼 sw	Timer 1 Image: Constraint of the second	3
Search	2.305 VCR ACTIVATION/DEACTIVATION	
ODE Libraries	COMMS/1E - ALL/PRE 4 12 MAY 06	
- COMMS	Show Ground Blocks	
P Nominal 2 305 - VCRs Activation/Deactivation	OBJECTIVE	
2.307 – VCR Record/Playback	The purpose of the procedure is activating and deactivating the VCRs.	
← ECLSS		л
∽ SIVQ	1. <u>VCRT ACTIVATION</u>	41
	1.1. Verify PDU1 Power Outlet	
	PWS EPDS: PDU1 28V Suspsys	
	PDU1 28V Susbsys	
	Venty Subsys PWP Bus4 - ON	
	[GROUND BLOCK HIDDEN]	
	1.2. Activate the VCR1	
	PWS COMMS: VCR1	
	(<u>vvri</u> 'Commands'	
	Verify Pwr - OFF	
	NOTE	
	VCP1 Activation ELAP performs: Power on VCP1 enable monitoring activate acquisition from VCP	
	1, switch VCR 1 to nominal mode, check that recorder is in remote mode.	
		-
Activating /col_data/odf/COMMS/NOMINAL/2.3		_
14Jun07.09:42:44 ERROR	I Equipment lext	-
14Jun07 09:42:42 INFO	Laptop3 Opened procedure /col_data/odf/COMMS/NOMINAL/2.305_E_COMMS_2305.xml	
14Jun07 09:42:41 ERROR	Laptop3 Display /col_data/displays/SYNOPTICS/MONITORING/MON_5.uss frozen	_
14/Jun07 / 93/42/29 EKKOK Laptops Display /col/data/displays/SYNOPTICS_ROOT/COL_HOME uss frozen		
14JUII07 03.40.30 ERROR	Laptops pispiay /col_data/uispiays/sinterines_recor/col_nome.tiss integen	

A Check List Procedure In The Procedure Executor

9.2.2.1 Working With Check List Procedure

Check List Procedures are executed in a different way. At first, a checklist contains parameters which have to be set before starting work. These parameters change the **Check List Procedure** depending on the circumstances in which it should be executed (i.e. a procedure could be used to configure a certain part of the equipment, but be applicable for more than one component. In this case the parameters could be used to select the desired equipment).

While executing a **Check List Procedure** the user performs the manual tasks described in the procedural sub steps of each step, one after another. The first step should be selected using the context menu and selecting the first step from the sub menu **Step**. The following single tasks are selected with the keyboard's arrow keys.

There are a few situations where the **Check List Procedures** might contain a decision and therefore more than one path to follow. These jumps are realized using **Step References**, which are links within the procedures text. A mouse click on these elements selects the according step of the procedure.

Another type of "link" are **Procedure Departures**. If a procedure requires the execution of another procedure (i.e. for initialization purposes) it will contain such a link to another procedure.

Check List Procedures may require the recording of specific data. This technique is called Data Recording and makes it possible to enter the data directly into the procedure.

A detailed description of this features can be found in the next section.

9.2.2.1.1 Elements In Checklist Procedures

Show Ground Blocks Button

This button will hide / show all procedure steps that are related to ground operations.

Hide Parameters Button

This button will hide / show the Parameter Selection.

Parameter Selection

Before the execution of a Check List Procedure the user should specify its parameters. This means he has to select the correct parameter using one of the radio buttons below the title of the procedure.

Depending on which parameter is selected, the procedure will be changed. That does not mean that the instructions or the work flow will change. The parameters have influence on certain descriptions or labels.



Check List Procedure - Parameter Selection

Instructions

Instructions are procedural steps which are textually described and have to be performed manually by the user on the according equipment. Instructions can be selected by using the arrow keys but contain no other functionality.

> Verify TCV CTCU Cntl Stat – CTCU1/2 A Selected Instruction

• Procedure Departure

In some cases the execution of another procedure may be necessary or desired. In this case a **Check** List Procedure may contain a link to another procedure. These links are called Procedure Departures. Clicking onto a Procedure Departure will immediately open the referenced procedure.

IMPORTANT

Read the instruction besides the **Procedure Departure** carefully as it may contain information about the step at which the current procedure has to be resumed after finishing the referenced one.

> 2.1.2. Perform X.1.XXX COL DECK RACK 1 ROTATE Steps 1 to 6 only EODF S&M / 1E-ALL / DRAFT 0 Continue with step 2.4

A Procedure Departure

• Step References

Although **Check List Procedures** have a linear work flow there may be some cases where a step has to be left undone and resuming at a later point in the procedure is necessary. Those non linear jumps in the work flow are realized by Step References. These references look similar to the Procedure Departures.

This text includes step 2 that st ep link and a and a reference li nk to Big Table

A Step Reference

• User Input Fields

There are certain procedures that ask the user to record some data. This can be done directly in the Procedure Executor using User Input Fields.

The **User Input Field** has to be selected to enable user input. After the user has entered the data he has to press the **Record** button to make the procedure store the entered data. After pressing the **Record** button it turns gray and changes its caption to **Recorded**.



It is possible to re-enter data after pressing the **Record** button. If the user has accidentally entered wrong data and pressed the **Record** button he may simply re-enter the correct data. The **Recorded** button will change back into the **Record** button as the input changes.

√TCV Cntl Stat – ENABLED					
Record Cabin Temp Setpoint : 72	Record				
Record Avg Cabin Temp :	Record				
[GROUND BLOCK HIDDEN]					
Data Recording In User Input Fields					
√TCV Cntl Stat – ENABLED					
Record Cabin Temp Setpoint : 72	(Recorded)				
Record Avg Cabin Temp : 69	Record				
[GROUND BLOCK HIDDEN]					

User Input Field After Recording a Value

• User Display Fields

The previously entered values may appear in the procedure at some later point. These areas are called **User Display Fields**.

User Display Fields are read only and just for verification.





• Clarifying Information

Among a procedure important issues are shown in special sections called **Clarifying Information**. These blocks are non functional but can be selected. Read this blocks carefully as they contain important information. There are three different types of Clarifying Information:

– Warning Blocks



Note Blocks

Caution Blocks

NOTE
1. First numbered caution with ' A Predef Message ' .
2. Second numbered caution with a MCC-H .

Note Block

9.2.2.1.2 Executing Check List Procedures The procedural steps of a Check List Procedure have to be executed one after another and there are no (or only a few) decisions that lead to different paths. Because of that the work flow is different from the one used in Logic Flow Procedures.

Check List Procedures are divided into a number of steps, which can be selected from the context menu. Each of this step contains selectable tasks, which can be selected by typing the arrow keys on the keyboard.

IMPORTANT

Only the main steps of each **Check List Procedure** can be selected by context menu. To select the sub steps within each step the arrow keys have to be used.

The selected sub step is displayed with a light green background. It contains the manual steps the user has to perform, and can range about a single or more lines.

9.2.3 PDF Procedures

PDF Procedures are procedures which are in the Adobe Acrobat PDF file format. To display a **PDF Procedure** the Adobe Acrobat Reader will be opened as a **Plugin** within lapap. It is not necessary to open the procedure externally.

PDF Procedures can be either Check List Procedures or Logic Flow Procedures.

Procedures of this type can only be viewed and executed manually. None of the functionality described above applies to **PDF Procedures**. Instead the functionality of the Adobe Acrobat Reader can be used (to print the procedure or to search for certain terms).

9.2.4 General Procedure Work Flow

There are some things that **Check List Procedures** and **Logic Flow Procedures** have in common:

• Interrupting The Execution

If a procedure has been executed until a certain step, and the user has to interrupt the execution, it is possible to stop the work and continue later. The work on the procedure may simply be stopped (by selecting another procedure, by closing the application or by choosing a different workspace). If the same procedure is opened the next time, a dialog box will ask the user if he wants to continue the execution of the procedure. If he chooses **Yes**, the procedure will be opened and scrolled to the position it had been stopped at. All values the user has entered into **User Input Fields** will be restored and the last step will be reselected.

If the procedure should be restarted completely, the user has to select **No** from the dialog box.

• Procedure Context Menu

The context menu of procedures contains the same element for both types of procedures.

• Printing Procedures

It is possible to print the currently opened procedure by selecting **Print** from the context menu. This will open the procedure as an external Adobe Acrobat Reader document (PDF). After this, the procedure may be printed using the Adobe Acrobat Reader.

9.3 Documentation Viewer

The **Documentation Viewer** may be used to view Documentation about procedures or general information concerning Columbus.

To open the **Documentation Viewer**, select the workspace **Documentation**. This will open the **Documentation Viewer** and load the documentations start page.

🛲 lapap 🍥		X
<u>File View Display Documentation Options</u>	Help	
Cesa 🔏 Documentation (R) 💌	2) 🖷 🔲 💭 🛊 🏠 🍖 🔍 🍕 🏘 🛶 🕴 🌩 🔹 Display Refresh 🛛 GPS 14Jun07 09:44:	41
💻 🍕 🕖 🖇 🔋 🕼 sw	Timer 1 Timer 2 Timer 3 Timer 4 00:00:00 ⁺ / ₁	3
<pre><enter search="" terms=""> Search</enter></pre>	Back Forward	
Procedures O Documentation	Crumb Trail: <u>RefDoc</u> > LAPAP Doc Plugin Test	
O Displays All	A	
Search: 🔾 Titles 🔾 Contents 🖲 Both		
Search Results shown here	Documentation	
	This index provides links to information on the following topics:	
	Reference Documentation:	
	 Help Crew Event Help Lapap Mk II User Manual 	
Time Classification Subsyste	n Equipment Text	
14Jun07 09:42:44 ERROR	Laptop3 Display /col_data/displays/SYNOPTICS/MONITORING/MON_3uss frozen	$- \ $
14Jun07 09:42:41 ERROR	Laptops opened procedure / Cul, DdddyDurg/Comm3/NOMINAL/2, 3005COMm3_2305.xmi	
14Jun07 09:42:29 ERROR	Laptop3 Display /col_data/displays/SYNOPTICS_ROOT/COL_HOME uss frozen	
14Jun07 09:40:38 ERROR	Laptop3 Display /col_data/displays/SYNOPTICS_ROOT/COL_HOME.uss frozen	

Documentation Viewer (Start Page)

The **Documentation Viewer** contains the following elements:

• Navigation Buttons

At the top of the **Workspace** are two navigation buttons labeled **Back** and **Forward**. These buttons work similar to the navigation buttons of an Internet browser. They navigate back and forward through the document history of the **Documentation Viewer**.

• Crumb Trail

The crumb trail below the navigation buttons displays the user's position within the hierarchical structure of the documentation. The single elements can be considered as branches within the tree structure of the documentation and can be clicked to open the corresponding page.

• Documentation View

The main area of the documentation Workspace shows the currently loaded page.

The documentation pages are of the format HTML or PDF. If a PDF documentation page is selected, the Adobe Acrobat Reader will be opened embedded in the **Documentation Viewer**.

The text size of HTML documentation can be changed by using the toolbar buttons or according menu elements.

9.3.1 Searching For Documentation

The contents of the documentation can be searched using the **ODF Search Plugin**. For detailed information on how to use the ODF Search Plugin, see Section 9.1.2.

Appendix A

Keyboard Shortcuts and Hot Keys

Alt + F3	Activate window menu
Alt + F4	Close active window
Alt + F7	Move active window
Alt + F8	Resize active window
Alt + F9	Minimize active window
Alt + F10	Maximize active window
Alt + Tab	Activate next window
Alt + Shift + Tab	Activate previous window
Alt + F1	Popup launch menu
Alt + F5	Show window list
Alt + F12	Mouse emulation using \rightarrow , \leftarrow , \downarrow and \uparrow
Alt + Print	Window screenshot
Ctrl + Print	Desktop screenshot
Ctrl + Alt + Delete	Logout
Ctrl + Escape	Show task manager
Ctrl + Alt + V	Show Klipper popup-menu
Ctrl + Alt + R	Manually invoke action on current clipboard
Ctrl + Alt + X	Enable/Disable clipboard actions
Ctrl + Alt + D	Toggle desktop
F1	User Manual
F2	Onboard Documentation
Ctrl + 1	Switch to Workspace "Synoptics"
Ctrl + 2	Switch to Workspace "Synoptics + Procedures"
Ctrl + 3	Switch to Workspace "Procedures"
Ctrl + 4	Switch to Workspace "CLSW Status"
Ctrl + 5	Switch to Workspace "Documentation"
F10	Activate application menu
F12	Columbus Home Page
	0
Tab	Focus next component
Shift + Tab	Focus previous component
	Perform dialog default action (normally Accept
Enter	/ OK / Open)
Space	Activate selected element
Esc	Dismiss dialog or cancel selection
	0

Table A.1: Lapap Mk II Application and Desktop Hot Keys
Alt + <alphanum.></alphanum.>	Activate item according to mnemonics (e.g.
	Transfer selection to next horizontal
\rightarrow	component or scroll contents right
←	Transfer selection to previous horizontal
	component or scroll contents left
\downarrow	Transfer selection to next vertical component or
	scroll contents down
\uparrow	Transfer selection to previous vertical
	component or scroll contents up
Home	Transfer selection to first component
End	Transfer selection to last component
	I I I I I I I I I I I I I I I I I I I
Ctrl + W	Close synoptic display
Ctrl + F4	Close synoptic display
Ctrl + U	Undock synoptic display
Ctrl + D	Dock synoptic display
Ctrl + F	Find text as you type
Ctrl + Shift + F	Find parameter in displays
Ctrl + ↑	Transfer selection to next section in checklist
	procedure
Ctrl + ↓	Transfer selection to previous section in
	checklist procedure
Y / y	'Yes' in logic flow procedure
N/n	'No' in logic flow procedure
X / x	'Next' in logic flow procedure
Ctrl + Backspace	Undo last decision in logic flow procedure

Table A.1: (continued)

Appendix **B**

Known Problems

• "Time synchronization occured" event messages (SPR-20091)

The cause for this message is generally a detected offset between the laptop local clock and the DMS provided station time. Usually the PWS laptop is synchronized to the DMS provided time regularly as long as it is connected to the LAN. In case the laptop is not connected to the LAN for a certain period but still running the laptop local clock may drift from the stable GPS time due to the inaccuracy of the laptop hardware. The same can happen if a process on the laptop blocks other tasks. In such situations the detected difference will result in the described message.

There are two situations where the message is raised although no time adjustment has been performed: Lapap start-up and shutdown.

• Using the CLSW Web Interface (SPR-20116)

The built-in web interface to the Columbus LAN Switch does not integrate smoothly. Especially in case new windows are opened interaction may be corrupted. E.g. when acknowledging CLSW events in the web interface a new window is opened with confirmation buttons. If in such situations the expected action is not performed a simple repetition of the required steps is adviced. Please note, that the steps may need to be repeated more than once.

• Hot Keys in the CLSW Web Interface (SPR-20683)

In the "CLSW Status" workspace the Lapap Mk II hot keys will not be available as long as the CLSW web interface has the input focus. Clicking with the mouse into the application's toolbar or status bar or switching into another workspace using the mouse will recover hot key functionality again.

Index

Α

Acknowledge Acknowledge Situational Awareness, 45 Acknowledging SAW Messages, 48 Action Block Element in Logic Flow Procedures, 57 Alarm Clock Timer Timer Types, 51 Audio Audio Notification Preferences, 18 С Caution Block Element in Check List Procedures, 62 Check List Procedure ODF Procedures following a linear work flow, 60 **Clarifying Information** Element in Check List Procedures, 62 Classification Message Attribute Classification, 42 Clear Removing all messages in the Message Buffer View, 45 Close Closing the Message Buffer View, 46 Combined Action / Decision Block Element in Logic Flow Procedures, 58 **Command Buttons** Command Buttons and 2 Step Commanding, 28 Context Menus

Context Menus of the different elements, 29 Count Down Timer Timer Types, 52

D

Data pool Save Data pool Buffer, 5 Decision Block Element in Logic Flow Procedures, 58 DMS Status DMS Status Display in Lapap tool bar, 40 Documentation Find in page..., 8 **Documentation Viewer** Viewing Documentation, 63

E

Element Types Element Types in Synoptic Displays, 28 Equipment Message Attribute Equipment, 42 **Execute Button** Button to execute commands in 2 Step Commanding, 28 Exit

Exit the application, 5

F

File File Menu, 5 Filter Filter System Messages, 43 **Find Parameters** Find Parameter in Other Display, 32 Freeze Freezing the Message Buffer view, 45

Η

Handling SAW messages Handling Situational Awareness, 48 Help Lapap Mk II Help, 8 Home Page Close all but Homepage, 7 Load Synoptics Home Page, 7

Ι Install

Installing Lapap Mk II on Linux/Unix, 3 Installing Lapap Mk II on Windows XP, 3 Instruction Block Element in Logic Flow Procedures, 57 Instructions Element in Check List Procedures, 61 International Procedure Viewer The IPV is a set of components to display and execute ODF Procedures, 53 IPV IPV = International Procedure Viewer, 53

L

Limit Manager The Limit Manager offers a way to change a Measurements soft limits / expected value, 35 Line Graph Line Graph Context Menu, 32 Line Graphs Line Graphs, 32 Live Setting the Message Buffer View back to live mode, 45 Load Loading System Messages, 45 Log Applications user Log, 6 Logic Flow Procedure ODF Procedures with a non linear work flow., 56 Μ

69

Measurements

Measurements, 29 Menu Menu Structure, 5 Message Buffer Message Buffer View, 41 Show the Message Buffer View, 5

Ν

Navigation Synoptic Display Navigation, 25 Navigation Buttons Synoptic Navigation Buttons, 27 Note Block Element in Check List Procedures, 62 Notes Element in Logic Flow Procedures, 58

0

ODF Book Browser A tree view to search and select ODF Procedures., 53 **Ops Products** Operational products in the standalone version, 4 **Overview Panel** Panel that gives an overview about the currently loaded Logic Flow Procedure, 57 Р Parameter Selection Setting the parameters for Check List Procedures, 61 Preferences Options Menu, 8 Preferences Dialog, 18 **Procedure Departure** Element in Check List Procedures, 61 Procedure Executor IPV Component in which ODF Procedures are

executed., 56 Procedure Panel Control in which Logic Flow Procedures are displayed and executed, 57 Procedure Properties Properties of ODF Procedures, 56

R

S

Record Button Element in Check List Procedures, 61 Recorded Button Element in Check List Procedures, 61 Requirements Check Platform Requirements, 3 Reset Application Reset Application, Read only Workspaces will be reverted, 5 Result Block Element in Logic Flow Procedures, 58

S-/KU-Band Communication Link Status Display in Lapap tool bar, 40 Save Saving System Messages, 45 SAW Situational Awareness, 47 SAW Icons Situational Awareness Icons, 47 SAW Message Buffer View Opening the Message Buffer View for Situational Awareness, 47 SAW messages Trigger Messages for Situational Awareness, 47 Search Searching for ODF Procedures, 54 Sort Sorting the Message Buffer View, 43 Station Mode Station Mode Status Display in Lapap tool bar, 39 Station Mode Compatibility Station Mode Compatibility in Lapap tool bar, 40 Status Displays Status Displays in Lapap tool bar, 39 Step Exits Element in Logic Flow Procedures, 59 Step Reference Element in Check List Procedures, 61 Subsystem Message Attribute Subsystem, 42 Open Subsystem Overview, 7 Sync Synchronizing the Message Buffer View, 45 Т Tabbed Mode Toggle Tabbed Mode, 7 Text Message Attribute Text, 42 U User Display Field Element in Check List Procedures, 62 User Input Field Element in Check List Procedures, 61 V View View Menu, 5 W

Warning Block Element in Check List Procedures, 62 Workspace Workspace Menu, 6 Workspaces Workspace Handling, 10